

Article A Novel Approach to Decision-Making on Diagnosing Oncological Diseases Using Machine Learning Classifiers Based on Datasets Combining Known and/or New Generated Features of a Different Nature

Liliya A. Demidova 匝

check for **updates**

Citation: Demidova, L.A. A Novel Approach to Decision-Making on Diagnosing Oncological Diseases Using Machine Learning Classifiers Based on Datasets Combining Known and/or New Generated Features of a Different Nature. *Mathematics* **2023**, *11*, 792. https:// doi.org/10.3390/math11040792

Academic Editor: Vladimir Balan

Received: 6 January 2023 Revised: 28 January 2023 Accepted: 2 February 2023 Published: 4 February 2023 Corrected: 4 May 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Institute of Information Technologies, Federal State Budget Educational Institution of Higher Education, MIREA—Russian Technological University, 78, Vernadsky Avenue, 119454 Moscow, Russia; liliya.demidova@rambler.ru

Abstract: This paper deals with the problem of diagnosing oncological diseases based on blood protein markers. The goal of the study is to develop a novel approach in decision-making on diagnosing oncological diseases based on blood protein markers by generating datasets that include various combinations of features: both known features corresponding to blood protein markers and new features generated with the help of mathematical tools, particularly with the involvement of the non-linear dimensionality reduction algorithm UMAP, formulas for various entropies and fractal dimensions. These datasets were used to develop a group of multiclass kNN and SVM classifiers using oversampling algorithms to solve the problem of class imbalance in the dataset, which is typical for medical diagnostics problems. The results of the experimental studies confirmed the feasibility of using the UMAP algorithm and approximation entropy, as well as Katz and Higuchi fractal dimensions to generate new features based on blood protein markers. Various combinations of these features can be used to expand the set of features from the original dataset in order to improve the quality of the received classification solutions for diagnosing oncological diseases. The best kNN and SVM classifiers were developed based on the original dataset augmented respectively with a feature based on the approximation entropy and features based on the UMAP algorithm and the approximation entropy. At the same time, the average values of the metric $MacroF_1$ -score used to assess the quality of classifiers during cross-validation increased by 16.138% and 4.219%, respectively, compared to the average values of this metric in the case when the original dataset was used in the development of classifiers of the same name.

Keywords: decision-making; oncological disease; kNN classifier; SVM classifier; dataset; features; UMAP algorithm; entropy; fractal dimension

MSC: 68Q32; 68T05

1. Introduction

Recently, elements of digital transformation have become increasingly visible and in demand in various areas of human activity, including healthcare.

The digital transformation of healthcare is a continuous process aimed at completely restructuring the mechanisms of work of industry authorities, medical organizations and their interaction with patients. The introduction of advanced digital technologies ensures high standards of medical care and the transition to the "4P medicine" model (preventive, personalized, participatory, predictive medicine) [1].

The introduction of digital technologies in healthcare should ensure a decrease in the level of morbidity and mortality of the population and an increase in life expectancy, including active life expectancy. The use of health monitoring technologies should allow not only for the detection of pathologies at an early stage, but also to prevent the development of diseases.

The development of technologies for analyzing large volumes of medical data, including the use of artificial intelligence, will make it possible to obtain new knowledge in the field of medicine and biology, as well as to develop new methods for diagnosing and treating diseases.

The transformation of healthcare under the influence of digital technologies is taking place everywhere, including such areas as:

- Transition from standardized clinical protocols to a personalized approach to patient care due to the accumulation of a large amount of medical data, as well as the widespread use of individual biomonitoring devices;
- Disease prevention through early diagnosis and regular health monitoring using wearable devices;
- Patient focus and active involvement of the patient in the treatment process.

At the same time, the demand for applied and computational mathematics' tools for digital environments becomes obvious in the case of solving problems of disease prevention through early diagnosis, including cancer [2–5].

Oncological diseases (ODs) are among the most dangerous ones because they can lead to serious consequences for patients, especially in the case of late diagnosis. Such consequences include significant pain and a difficult psychological state. The treatment of oncological diseases is usually lengthy and involves significant financial costs both for the patients and their families and for the state.

An OD is an immune disease that first causes the division and growth of abnormal cells in a single organ of the patient, and then can quickly spread to the entire body.

Obviously, early diagnosis of an OD should allow the oncologist to choose an adequate and effective treatment regimen for the patient in a timely manner. However, the task of early diagnosis of an OD is very difficult: unfortunately, the typical symptoms of ODs appear only in the later stages when the disease is difficult to treat, so the treatment may ultimately turn out to be unsuccessful.

One of the approaches to the early diagnosis of an OD is based on the analysis of test results, in particular, gene tests (GTs) [4,6] and protein tests (PTs) [4,7,8]. GTs reveal hereditary information, they are static, difficult to interpret, and are usually used to detect congenital genetic diseases [4]. Such tests do not detect diseases that occur when there are problems with the immune system and metabolism. In addition, GTs are expensive. PTs, unlike GTs, are dynamic and they allow (if carried out in a timely manner) for identifying the occurrence of an OD and track its development [4]. Additionally, PTs are non-invasive, painless for patients and affordable.

The diagnosis of ODs based on blood protein markers has been increasingly used in the last few years [4,9,10]. Presumably, PT-based diagnostic technologies will allow for predicting the risks of developing oncological diseases 1–3 years in advance, which will make it possible to take advanced preventive measures against the emergence and development of an OD.

There are many different types of protein markers in the blood, and the values of the markers are different for different types of ODs [4]. It is obvious that the use of only one type of marker or some limited group of markers reduces the accuracy of diagnosing ODs, even if this marker (or group of markers) is recognized as the standard for diagnosis. One of the reasons leading to the refusal to use the full range of PTs results is the limited ability of specialists to interpret the data. It is easier to operate with information about the values of 1–2 protein markers than with data on the values of several dozen of the same markers (especially if specialists use a unified table with a priori given limits of the norm and abnormality, followed by a decision based on their knowledge and intuition). It can be reasonably assumed that the use of the full range of PTs results should improve the accuracy of diagnosing various ODs.

It is obvious that the involvement of modern technologies of data mining and machine learning will make it possible to reveal the knowledge about the relationship between the values of blood protein markers in various types of ODs hidden in the PT results.

Currently, data mining and machine learning technologies are actively used to solve various problems of medical diagnostics, including solving the problems of diagnosing ODs [10–19]. In this case, classifiers are developed based on appropriate datasets. The k-Nearest Neighbors (kNN) algorithm [10], SVM (Support Vector Machine) algorithm [11], Decision Tree (DT) algorithm [12], Random Forest (RF) algorithm [15], their hybrids and deep learning algorithms [17–19] are most broadly used in creating such classifiers.

The main problem in the development of tools for solving problems of medical diagnostics is the imbalance of classes in the dataset [10,20]: usually, normal data patterns characterizing cases when the disease is absent constitute the majority class, while data patterns of interest and characterizing cases when certain diseases are present constitute the corresponding minority classes. In this case, for example, instead of developing a binary classifier based on an imbalanced dataset, it becomes necessary to develop a multiclass classifier based on an imbalanced dataset, which is a non-trivial task, since it is necessary to teach the classifier to separate data from different classes that are imbalanced. Obviously, a binary classifier can be obtained even in the case of class imbalance (for example, in a ratio of 10:1), and the accuracy of it will be 90% (which is not bad in the case of balanced classes), but such accuracy should be considered bad if the classifier made errors on all objects from the minority class.

Data scientists have proposed various approaches to solve the problem of class imbalance:

- Approaches using various class balancing algorithms that implement oversampling technologies (for example, SMOTE algorithm (Synthetic Minority Oversampling Technique) [21–23], ADASYN algorithm (Adaptive Synthetic Sampling Approach) [24]), undersampling technologies (for example, Tomek Links algorithm) [23,25] and their combinations;
- Approaches that apply algorithms that account for the sensitivity to the cost of wrong decisions (cost-sensitive algorithms) [10];
- Approaches that implement the transfer of data into a space of a new dimension (with a decrease [26–32] or increase [11,33,34] in dimension), in which data classes will be separated from each other better than in the original space;
- Approaches that implement the so-called one-class classification [34–37].

It should be noted that the transition to the space of a new dimension can be implemented in various ways, for example, using:

- Dimension reduction algorithms (linear [26,27] and non-linear [28–32]) that allow one to move to a space of lower dimension;
- Kernel functions (as it is done in the SVM-algorithm, which allows the transition to a space of higher dimension) [11,33,34,36];
- Algorithms for engineering (generation) of new features for data patterns, which make it possible to move to a space of higher dimension [38].

All of these approaches have their advantages and disadvantages, and there is no universal methodology for choosing the approach that is appropriate to apply to the dataset used in the development of the data classifier.

In each case, it is necessary to perform a comprehensive analysis in order to:

- Avoid loss of information during undersampling or reducing the dimension of the data space;
- Exclude the introduction of false or redundant information during oversampling or increasing the dimension of the data space.

One of the obvious tools that can be used in assessing the quality of the developed classifier is detailed analysis of various classification quality metrics on the test sample, including metrics that allow for accounting for the specifics of the dataset, namely, its imbalance (for example, it is appropriate to use metrics such as *Balanced accuracy* and F_1 –*score*).

In addition, it is advisable to use cross-validation, including k-fold cross-validation, which makes it possible to empirically evaluate the generalizing ability of the developed classifier.

A large number of papers have been devoted to the problem of analyzing medical datasets containing information about blood protein markers and making decisions on the diagnosis of ODs. In particular, in the pilot study [4], the task was to identify and surgically localize eight types of resectable ODs (ovary, liver, stomach, pancreas, esophagus, colorectum, lung and breast) based on a multi-analysis blood test using the CancerSEEK test [4,5] based on a machine learning algorithm called the logistic regression algorithm. The dataset used in the study is publicly available and located in the Catalog of Somatic Mutations in Cancer (COSMIC) repository [39] as NIHMS982921-supplement-Tables_S1_to_S11.xlsx, and it is constantly updated with new data. The original dataset is available in the Supplementary Material for the paper as aar3247_cohen_sm_tables-s1-s11.xlsx [4]. The eight types of ODs considered in the dataset were chosen by the authors because they are the most common among the population of Western countries, and also because blood-based tests are not used in clinical practice for their early detection.

The authors of the pilot study [4] proposed to assess the levels of circulating proteins and mutations in cell-free DNA and use the obtained data in the LC (logistic classifier)based CancerSEEK test. The study used 1005 data patterns received from patients with non-metastatic clinically identified ovarian, liver, stomach, pancreas, esophagus, colon, lung or breast ODs. The CancerSEEK test was based on screening results, such as the levels of protein biomarkers and ctDNA (Circulating tumor DNA). The authors proposed to take into account each person's gender, the levels of eight proteins and the presence of mutations in 1933 different genomic positions, each of which can mutate in several ways.

It was assumed that the presence of a mutation in the analyzed gene or an increase in the level of any of the eight proteins makes it possible to classify the patient's pattern as positive, i.e., a pattern with an identified OD. The authors used logarithmic ratios to evaluate the mutations and included them in a logistic regression algorithm (and, accordingly, in the LC), which took into account both the mutation data and the protein biomarker levels to evaluate the results of the CancerSEEK test. At the same time, the average values of sensitivity and specificity were determined from the results of 10 iterations using 10-fold cross validation.

The results of the experiments in [4] showed that the average value of the *Sensitivity* metric for eight types of ODs is about 70%, the highest value of the sensitivity metric is achieved for the ovary class (98%), slightly less for the liver class (close to 98%), and the lowest value of the sensitivity metric is achieved for the breast class (33%). In general, the values of the *Sensitivity* metric ranged from 69 to 98% in the detection of five types of ODs (esophagus, pancreas, stomach, liver and ovaries, listed in ascending order of values of the *Sensitivity* metric). The value of the metric specificity was above 99%: only 7 of the 812 patterns without known cancers scored positive. At the same time, as the authors write, there is no certainty that several false-positive patterns of patients identified in the normal class do not actually have an undiscovered OD. However, the classification of these patterns as false positives can be considered the most conservative approach to the classification and interpretation of medical data.

The average sensitivity of the CancerSEEK test was 73% for stage II cancer, which is the most common, 78% for stage III cancer, and less than 43% for stage I cancer. The highest and the lowest sensitivity for stage I cancer (the earliest stage) was recorded for liver cancer (100%) and esophageal cancer (20%), respectively.

A liquid biopsy involves deriving mutant DNA templates from dying cancer cells. Later on, these templates serve as specific neoplasia markers. The authors of the pilot study [4] found that the mutation in blood plasma was identical to the mutation found in the primary tumor of the same person in 90% of analyzed cases diagnosed with ODs. This correspondence between the plasma and primary tumor was evident for all eight types of cancer and ranged from 100% for ovarian and pancreatic cancer to 82% for gastric

cancer. One disadvantage of liquid biopsies is the inability to determine the type of cancer in patients with positive tests, leading to clinical problems for follow-up.

The authors of the pilot study [4] point out that the vast majority of information about the localization of ODs was obtained from protein markers, since mutations in the driver gene are usually not tissue-specific. They propose the LC that implements the CancerSEEK test for the presence of eight common types of ODs. During the development of the LC, the authors used a training set that combined information about the levels of the gene and protein biomarkers. In doing so, they were able to increase the sensitivity of the CancerSEEK test without a significant decrease in specificity. The authors note that the effectiveness of combining completely different agents with different mechanisms of action is widely recognized in therapy, but not usually used in diagnostics. They also believe that other cancer biomarkers, such as metabolites, mRNA transcripts, microRNAs or methylated DNA sequences, can be similarly combined to increase the sensitivity and localization of a cancer focus.

The followers of the authors of the pilot study continued to offer their own versions of classifiers that implement the diagnosis of ODs based on datasets on values of blood protein markers without taking gene markers into account.

For example, in [10], the authors proposed to use a cost-sensitive kNN algorithm based on a three-class imbalanced dataset containing patterns characterized by 39 blood protein markers and belonging to one of three classes: normal, ovary and liver. In order to include the hidden information in the data analysis and improve the quality of the classification, the authors used two entropy metrics: approximate entropy AE and sample entropy SE. At the time that the study was conducted, the analyzed dataset contained 897 patterns from three classes in the ratio Normal:Ovary:Liver = 799:54:44. The overall accuracy of the classifier was equal to 0.952, and the values of such metrics as *Precision, Recall, MacroF*₁–*score* and *AUC* were equal to 0.807, 0.833, 0.819 and 0.920, respectively.

It should be noted that the authors of this work abandoned the attempt to develop a classifier for nine classes (eight classes corresponding to patterns of various types of ODs, and one class corresponding to patterns for which no ODs were diagnosed) due to poor separability of the classes.

The goal of this study is to develop the high-precision data classifiers on ODs using modern tools for data mining and machine learning technologies. It is supposed to develop kNN [10,34] and SVM [11,33,34,36] classifiers using oversampling algorithms SMOTE [21], Borderline SMOTE-1 [22], Borderline SMOTE-2 [22] and ADASYN [24], which allow for restoring the balance of classes in the original and extended datasets, formed on the base of the original dataset with the application of various techniques for extracting new features. In particular, it is planned to consider:

- The UMAP (Uniform Manifold Approximation and Projection) algorithm [29–32], which implements non-linear data dimensionality reduction by embedding data into a space of lower dimensionality to generate new features based on pattern features;
- Formulas for calculating entropies [40–45], Hjorth parameters [46,47] and fractal dimensions [48–52] in order to generate new features based on the pattern features.

This study presents the first attempt to generate datasets that involve different tools with the subsequent selection of the best of them to form new features by extracting information hidden in the features of the original dataset. The generation of new features was carried out using the UMAP algorithm, as well as various formulas for calculating entropies, Hjorth parameters and fractal dimensions. Various combinations of new features were selected based on correlation and the perceived ability to distinguish between patterns of different classes were added to the features of the original dataset or combined to form a new dataset. As a result, it becomes possible to work simultaneously with different datasets that describe the subject area. This approach to the generation of datasets is used for the first time in the field of medical diagnostics, including the field of diagnostics of oncological diseases based on blood protein markers. Subsequently, these datasets are balanced using oversampling algorithms and are used in the development of classifiers based on the kNN

and SVM algorithms with subsequent selection of the best classifiers based on classification quality metrics.

The rest of the paper is organized as follows. Section 2 briefly describes the aspects of developing the kNN and SVM classifiers. It also discusses the applied quality metrics of multiclass classification, as well as the problem of class imbalance in datasets. In addition, this section provides a summary of the principles of operation of the UMAP algorithm, as well as background information on the investigated nonlinear data extraction tools that are difficult to obtain from traditional statistics, in particular, on entropy characteristics and fractal dimensions. Section 3 presents a description of the novel approach to the formation of the datasets used in the development of the classifiers. Section 4 is devoted to the analysis of the original dataset based on the UMAP algorithm with data visualization, aspects of the generation of new features based on entropy characteristics and fractal dimensions of the data patterns, choosing the best of them. It also discusses all of the steps of the creation of new datasets used for the development of the kNN and SVM classifiers and the results of the development of such classifiers, accompanied by tables and figures. Section 5 presents a discussion of the proposed results. Finally, Section 6 contains conclusions and goals regarding future work.

2. Materials and Methods

2.1. Aspects of Development of Classifiers

The development of classifiers can be performed using various machine learning algorithms, for example, kNN [10,34,53–56], SVM [10,33,34,36,57,58], LR (Logistic Regression) [4,59], DT [12–14], RF [15,60,61] and neural networks [17–19], as well as cascade algorithms and ensembles based on them. We can use the default values of the classifier parameters, or fine-tune them using population optimization algorithms [62,63]. The quality of classification using such classifiers will depend both on the quality of datasets on the basis of which the classifiers are developed, and on the specifics of the mathematical apparatus embedded in the algorithms corresponding to the classifiers. Currently, there is no universal machine learning algorithm that could ensure that a classifier developed on its basis will provide high quality of classification for any arbitrary training dataset. The same can be said about cascade classifiers and ensembles of classifiers. Obviously, it is desirable to minimize the time for developing a classifier that provides high-quality data classification.

Two machine learning algorithms (kNN and SVM) are considered in this study, although all the ideas formulated below can be used in the development of classifiers based on any machine learning algorithm because they affect only the stage of preparation of datasets used in the development of classifiers.

The choice of the kNN algorithm is due to the simplicity of its implementation and, as a result, low time costs for the development of the kNN classifier.

The choice of the SVM algorithm is due to the availability of tools for working with various kernel functions that provide a transition to a higher-dimensional space, and the possibility of generating classification rules in an explicit form by identifying the socalled support vectors located along the class boundary. However, the time spent on the development of the SVM classifier becomes significantly larger compared to the time spent on the development of the kNN classifier.

It is precisely because of the large time costs for the development of the RF classifier that the proposed study does not consider the RF algorithm, although when it is used, it is usually possible to obtain classifiers that provide a very high quality of data classification.

Let $U = \{\langle x_1, y_1 \rangle, \dots, \langle x_s, y_s \rangle\}$ be a dataset used in the development of classifiers, where $x_i \in X$; $y_i \in Y = \{1, \dots, M\}$; $i = \overline{1, s}$; s is the number of patterns in the dataset; M is the number of classes; X is the set of signs of patterns; and Y is the set of pattern class labels [34].

Let classifiers be trained on *S* patterns and testing be carried out on s - S patterns. The quality of the classifiers is assessed using the k-fold cross-validation procedure.

2.1.1. kNN Classifier Development

The number of nearest neighbors k for the pattern x and the method of voting for k nearest neighbors on the question of the class membership of the pattern x are specified during the development of the binary kNN classifier. The values of the parameters should provide the minimum classification error.

The membership class y of a pattern x is determined by the membership class of most of the patterns from among the k nearest neighbors of the pattern x. Various metrics, such as Euclidean, cosine, Manhattan, etc. can be used to calculate the distance between patterns in the kNN algorithm. Usually, the distance is calculated using the Euclidean distance metric as [34,53,54]:

$$d(x_i, x) = \sqrt{\sum_{h=1}^{q} (x_i^h - x^h)^2},$$
(1)

where *q* is the number of features of patterns x_i ($i = \overline{1, k}$) and x; x_i^h is the *h*-th coordinate of *i*-th pattern x_i ; and x^h is the *h*-th coordinate of pattern x.

Various voting methods can be used to determine the membership class of a pattern during the development of the kNN classifier, for example, simple unweighted voting and weighted voting [34,53,54].

Simple unweighted voting works in such a way that the distance from the pattern *x* to each of the *k* nearest neighbors x_i ($i = \overline{1, k}$) does not matter: each of the *k* nearest neighbors x_i ($i = \overline{1, k}$) of the pattern votes for its assignment to its class, and all *k* nearest neighbors x_i ($i = \overline{1, k}$) have equal rights in class definition for pattern *x*. The pattern *x* will be assigned to the class that receives the most votes:

$$u = \operatorname*{argmax}_{m \in Y} \sum_{i=1}^{k} |y_{x_i, x} = m|, \qquad (2)$$

Weighted voting works in such a way that the distance from the pattern x to each of the k nearest neighbors x_i ($i = \overline{1, k}$) is taken into account: the smaller the distance $d(x_i, x)$, the more significant the contribution to the estimation of the pattern x belonging to a certain class is made by the vote of the nearest neighbor x_i ($i = \overline{1, k}$). The total contribution of the votes of the nearest neighbors x_i ($i = \overline{1, k}$) for the pattern x belonging to the class with the label $m \in Y$ can be calculated as:

$$\alpha_m = \sum_{i=1}^k \frac{1}{d^2(x_i, x)} \cdot r_{i,m},$$
(3)

where $r_{i,m} = 0$, if $y_{x_i,x} \neq m$ and $r_{i,m} = 1$, if $y_{x_i,x} = m$.

α

The class that received the highest value α_m according to Formula (3) is assigned to the pattern *x*.

The problem of finding the optimal values of the parameters of the kNN classifier, for example, the number of neighbors and the voting method, can be solved using grid search algorithms or evolutionary optimization algorithms, such as the genetic algorithm, differential evolution algorithm, PSO algorithm, bee algorithm, ant colony algorithm, fish school algorithm, etc.

2.1.2. SVM Classifier Development

The SVM algorithm implements binary classification [34,57,58]. In the case of multiclass classification, i.e., for $M \ge 3$, strategies such as OvO (One-vs-One) or OvR (One-vs-Rest) are used, which allow for the use of binary classification solutions to form solutions for multiclass classification [64].

The One-vs-One strategy breaks the multiclass classification into one binary classification problem for each pair of classes. The One-vs-Rest strategy breaks the multiclass classification into one binary classification problem for each class. In the case of binary classification, the basic SVM algorithm operates on patterns $x_i \in X$ with class labels $y_i \in Y = \{-1; +1\}$.

The development of the SVM classifier can be performed using kernel functions $\kappa(x_i, x_{\tau})$ such as linear, polynomial, radial basis and sigmoid kernel functions. Particularly, the linear kernel function has the form $\kappa(x_i, x_{\tau}) = x_i \cdot x_{\tau}$, and the radial basic kernel function, which is used in this study, has the form $\kappa(x_i, x_{\tau}) = exp(-(x_i - x_{\tau}) \cdot (x_i - x_{\tau})/(2 \cdot \sigma^2))$, where $x_i \cdot x_{\tau}$ is the scalar product for x_i and x_{τ} and σ ($\sigma > 0$) is the kernel function parameter.

The value of the regularization parameter C (C > 0) [57], the type of the kernel function $\kappa(x_i, x_\tau)$ and the values of the parameters of the kernel function (for example, the value of parameter σ for the radial basis kernel function) are determined during the development of the binary SVM classifier. The values of the parameters should provide the minimum classification error.

The development of the binary SVM classifier involves solving the problem of construction of a hyperplane separating the classes. According to the Kuhn–Tucker theorem, this problem can be reduced to a quadratic programming problem containing only dual variables λ_i ($i = \overline{1, S}$) [34,57,58]:

$$\begin{cases} \frac{1}{2} \cdot \sum_{i=1}^{S} \sum_{\tau=1}^{S} \lambda_{i} \cdot \lambda_{\tau} \cdot y_{i} \cdot y_{\tau} \cdot \kappa(x_{i}, x_{\tau}) - \sum_{i=1}^{S} \lambda_{i} \to \min_{\lambda}, \\ \sum_{i=1}^{S} \lambda_{i} \cdot y_{i} = 0, \\ 0 \le \lambda_{i} \le C, i = \overline{1, S}. \end{cases}$$
(4)

The support vectors are determined as a result of the training of the binary SVM classifier. These are feature vectors of learning patterns x_i , for which the values of the corresponding dual variables λ_i are not equal to zero ($\lambda_i \neq 0$) [57]. The support vectors carry all the information about class separation since they are located near the hyperplane separating them.

The classification decision rule that assigns a membership class to a pattern with the label "-1" or "+1" is defined as [34,57,58]:

$$F(z) = sign\left(\sum_{i=1}^{S} \lambda_i \cdot y_i \cdot \kappa(x_i, x) + b\right),$$
(5)

where $b = w \cdot x_i - y_i$; $w = \sum_{i=1}^{S} \lambda_i \cdot y_i \cdot x_i$.

The problem of finding the optimal values of the SVM classifier parameters, for example, the regularization parameter *C* (*C* > 0), the type of the kernel function $\kappa(x_i, x_\tau)$ and the values of the parameters of the kernel function, can be solved using grid search algorithms or evolutionary optimization algorithms.

2.2. Quality Metrics of Multiclass Classification

The assessment of the quality of a multiclass classification can be performed using metrics such as *Accuracy*, *MacroPrecision*, *MacroRecall* and *MacroF*₁–*score* [65].

The *accuracy* metric for multiclass classification can be calculated as:

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN},$$
(6)

where TP is the total number of true positive elements in the multiclass classification problem; TN is the total number of true negative elements in the multiclass classification problem; FP is the total number of false positive elements in the multiclass classification problem; and FN is the total number of false negative elements in the multiclass classification problem. The MacroPrecision metric for multiclass classification can be calculated as:

$$MacroPrecision = \frac{1}{M} \sum_{m=1}^{M} \left(\frac{TP_m}{TP_m + FP_m} \right), \tag{7}$$

where *m* is the class label index; TP_m is the number of true positive elements for the *m*-th class label; FP_m is the number of false positive elements for the *m*-th class label; and *M* is the total number of classes in the multiclass classification problem.

The *MacroRecall* metric for multiclass classification can be calculated as:

$$MacroRecall = \frac{1}{M} \sum_{m=1}^{M} \left(\frac{TP_m}{TP_m + FN_m} \right), \tag{8}$$

where *m* is the class label index; TP_m is the number of true positive elements for the *m*-th class label; FN_m is the number of false negative elements for the *m*-th class label; and *M* is the total number of classes in the multiclass classification problem.

The *MacroF*₁-*score* metric for multiclass classification can be calculated as:

$$MacroF_1 - score = 2 \cdot \left(\frac{MacroPrecision \cdot MacroRecall}{MacroPrecision + MacroRecall}\right),\tag{9}$$

Such metrics as (7)–(9) are useful and effective when developing classifiers using imbalanced datasets.

The $MacroF_1$ -score is used as a maximized classification quality metric. Based on *MacroPrecision* and *MacroRecall*, this metric makes it possible to simultaneously account for information on precision and recall of the solutions generated by the classifiers.

2.3. Solving the Class Imbalance Problem

In most machine learning algorithms, it is assumed that the goal of learning is to maximize the proportion of correct decisions in relation to all decisions made, and both the population and training dataset obey the same distribution. In this case, the datasets used to develop classifiers should be class-balanced, and the cost of a classification error for all patterns should be the same.

However, in many practical problems, one has to work with datasets that are poorly balanced. Learning on imbalanced datasets (imbalance problem) [66,67] can lead to a significant decrease in the quality of the classification solutions obtained using classifiers developed on their basis, since such datasets do not provide the required data distribution characteristics used in training.

For example, in the case of binary classification, the imbalance of the dataset is that more of the patterns in the dataset belong to one class, which is commonly called "majority", and a much smaller set of patterns belongs to another class, which is commonly called "minority".

The cost of misclassifying minority class patterns is often many times more expensive than misclassifying majority class patterns, because minority class objects in real datasets represent rare but most important instances.

For example, in the context of developing classifiers based on a dataset for ODs, the class of normal patterns is the majority, while the classes describing eight types of ODs are minor. Obviously, the correct classification of patterns of minority classes both for the problem under consideration and in general is of significant interest.

There are various approaches to solving the problem of class imbalance based on the application of:

- Class balancing algorithms that implement oversampling, undersampling and their combinations [21–25];
- Algorithms accounting for the sensitivity to the cost of wrong decisions [10];

- Algorithms for transferring data into a space of a new dimension (with a decrease or increase in dimension) in order to improve the separability of data [11,26–34];
- One-class classification algorithms [34–37].

In this study, the imbalance problem is proposed to be solved using such oversampling algorithms as SMOTE [21], Borderline-SMOTE-1 [22], Borderline-SMOTE-2 [22] and ADASYN [24], followed by choosing the best of them in the sense of providing a higher quality of data classification.

The use of oversampling algorithms is due to the fear of losing critical data patterns in the case of using undersampling algorithms in the context of the problem of developing classifiers for diagnosing oncological diseases.

The SMOTE (synthetic minority oversampling technique) algorithm [21] randomly selects a minority class pattern named *a* and randomly chooses a pattern named *b* from the *k* nearest neighbors of pattern *a*. The synthetic pattern will be located randomly in the line segment between the original pattern *a* and its neighbor *b*. The same process will be repeated until the desired number of synthetic patterns is reached.

The Borderline-SMOTE-1 algorithm [22] considers only the minority class patterns that have a number of minority class neighbors in the range [g/2, g], where g is defined by the researcher. These are the borderline patterns and they can be easily misclassified, i.e., they are "in danger". After detecting such original patterns, a basic SMOTE algorithm is applied to create synthetic patterns.

The Borderline-SMOTE-2 algorithm [22] is similar to the Borderline-SMOTE-1 algorithm, but it also considers the neighbors of the majority class.

The ADASYN algorithm [24] is similar to the SMOTE algorithm, since it generates synthetic patterns in the line segments between two minority class original patterns. It uses a weighted distribution for different minority class patterns that takes into account their level of difficulty: the minority class patterns that have fewer minority class neighbors are harder to learn than those which have more neighbors of the same class. The ADASYN algorithm generates more synthetic patterns for the minority class original patterns that are harder to learn and generates less for the minority class original patterns that are easier to learn.

2.4. UMAP Algorithm

The UMAP algorithm performs nonlinear dimensionality reduction preserving both local and global structures of high dimensional data in the best possible way [29,32] compared to other similar algorithms, for example, compared to the t-SNE algorithm [28].

The UMAP algorithm builds a fuzzy weighted undirected graph in the first step and optimizes loss function in the second step [29].

The UMAP algorithm works with a dataset $X = \{\vec{x}_1, \vec{x}_2, \dots, \vec{x}_s\}$, which contains *s* objects (patterns). Every object $\vec{x}_i \in X$ is represented by *q*-dimensional vector: $\forall \vec{x}_i \in \mathbb{R}^q$. The UMAP algorithm embeds objects from the *q*-dimensional space into the *h*-dimensional space ($h \le q$).

In the first step, the UMAP algorithm searches for the *k* nearest neighbors $Z_i = \{\vec{z}_{i1}, \dots, \vec{z}_{ik}, \dots, \vec{z}_{ik}\}$ for every object $\vec{x}_i \in X$, where $\forall \vec{z}_{il} \in X$; $i = \overline{1, s}$; $l = \overline{1, k}$, as described in [68]. Then it computes the scalar distance value d_{il} between \vec{x}_i and $\vec{z}_{il} \in Z_i$ using a distance metric. In the case of working with a Euclidean distance metric, the scalar value d_{il} can be calculated as (1), where pattern x_i is replaced by \vec{x}_i and x is replaced by \vec{z}_{il} . As a result, for each object $\vec{x}_i \in X$, the UMAP algorithm determines a set $D_i = \{d_{i1}, \dots, d_{ik}\}$, which contains the distances between \vec{x}_i and each of its *k* nearest neighbors.

After that, a fuzzy simplicial set, represented as a vector $\vec{\mu}_i \in \mathbb{R}^s$, is constructed for each object. First, the UMAP algorithm searches for $\rho_i \in D_i$, such that $\rho_i \leq d_{il}$ for

every $d_{il} \in D_i$. After that, a binary search is implemented in order to find σ_i satisfying the following condition:

$$\sum_{l=1}^{k} e^{\left(\frac{\rho_{i} - d_{il}}{\sigma_{i}}\right)} = \log_{2} k.$$
(10)

Then the *j*-th component of vector $\vec{\mu}_i$ is represented by a fuzzy value, which shows how similar *i*-th and *j*-th objects from the X set are. Therefore, if the two objects, \vec{x}_i and \vec{x}_j , are not neighbors, then the *j*-th component μ_{ij} of vector $\vec{\mu}_i$ is set to 0. If the two objects, \vec{x}_i and \vec{x}_j , are neighbors, then the *j*-th component μ_{ij} of vector $\vec{\mu}_i$ is computed as:

$$\mu_{ij} = e^{\left(\frac{\rho_i - d_{ij}}{\sigma_i}\right)}.$$
(11)

It is necessary to say that $\mu_{ij} \in [0, 1]$.

The UMAP algorithm defines a vector $\vec{\mu}_i \in \mathbb{R}^s$ for each object $\vec{x}_i \in X$. This vector encodes fuzzy similarities between the *i*-th object and every *j*-th object belonging to the original high dimensional dataset *X*.

As a result, the UMAP algorithm builds a weighted adjacency matrix $Matr \in \mathbb{R}^{s \times s}$, where each *i*-th row is represented by fuzzy vector $\vec{\mu}_i$ ($i = \overline{1, s}$).

The weighted adjacency matrix *Matr* represents a fuzzy weighted oriented graph, which codes pairwise similarities of objects from X. Matrix *Matr* is not symmetric.

In the second step, the asymmetric matrix *Matr* is symmetrized using probabilistic t-conorm:

$$\mu_{ij} \leftarrow \mu_{ij} + \mu_{ji} - \mu_{ij}\mu_{ji},\tag{12}$$

where *i* and *j* are the numbers of rows and columns in the *Matr* matrix, respectively $(\mu_{kk} = 0; k = \overline{1, s})$.

Thus, the transformed matrix *Matr* will already be symmetric.

The initial low dimensional representations of objects from the set *X* described by *q*-dimensional vectors in the space \mathbb{R}^h are calculated using spectral embedding [32] ($h \le q$). As a result, the matrix $Y \in \mathbb{R}^{s \times h}$ is obtained.

Then the UMAP algorithm starts the optimization process using the weighted fuzzy cross-entropy with reduced repulsion as the loss function [69]:

$$L(Matr, Y) = \sum_{i=1}^{s} \sum_{j=1}^{s} \left(\mu_{ij} \ln \frac{\mu_{ij}}{\nu_{ij}} + \frac{\sum_{k=1}^{s} \mu_{ik}}{2s} \ln \left(\frac{1 - \mu_{ij}}{1 - \nu_{ij}} \right) \right), \tag{13}$$

where $Matr \in \mathbb{R}^{s \times s}$ is the symmetric adjacency matrix containing fuzzy values encoding pairwise similarities of high dimensional objects from the dataset $X; Y \in \mathbb{R}^{s \times h}$ is the representation of *s* objects in the low dimensional space $\mathbb{R}^h; \mu_{ij} \in [0, 1]$ is a scalar value defining the fuzzy similarity of *i*-th and *j*-th high dimensional objects from the original dataset X; and $\nu_{ij} \in [0, 1]$ is a scalar value defining the fuzzy similarity of *i*-th and *j*-th objects in low dimensional space \mathbb{R}^h .

The UMAP algorithm determines the pairwise similarity v_{ij} of the *i*-th and *j*-th objects represented by the *i*-th and *j*-th rows of the $Y \in \mathbb{R}^{s \times h}$ matrix in the low dimensional space \mathbb{R}^h as:

$$\nu_{ij} = \left(1 + a d_{ij}^{2b}\right)^{-1},\tag{14}$$

where d_{ij} is the scalar value between the *i*-th and *j*-th objects, described by vectors \vec{y}_i and \vec{y}_i corresponding to the rows in the matrix *Y*.

The scalar value d_{ij} can be computed using a Euclidean distance metric (1), assuming vectors \vec{x}_i and \vec{z}_{il} in (1) are replaced by vectors \vec{y}_i and \vec{y}_i , respectively, and q is replaced

with *h*; *a* and *b* are the coefficients that are chosen by non-linear least squares fitting of (14) against the curve:

$$\psi_{ij} = \begin{cases} 1, d_{ij} \le d_{min} \\ e^{(d_{min} - d_{ij})}, d_{ij} > d_{min}' \end{cases}$$
(15)

where d_{ij} is the distance value between the *i*-th and *j*-th objects \vec{y}_i and \vec{y}_j represented by rows in the matrix Y; and d_{min} is the parameter of the UMAP algorithm ($d_{min} \in (0, 1]$), which affects the density of the clusters formed during the loss function (13) optimization process in the low dimensional space \mathbb{R}^h .

The UMAP algorithm performs the optimization of the loss function (13) using the stochastic gradient descent (SGD) algorithm [29]. The locations of objects, which are described by rows of the matrix $Y \in \mathbb{R}^{s \times h}$, are specified during each iteration of the SGD algorithm in order to minimize the loss function (13). Other functions given, particularly in [34], can also be used as a loss function.

2.5. Entropies, Hjorth Parameters and Fractal Dimensions

A number of works have shown that the use of nonlinear approaches can help extract some information from the data that is difficult to obtain from traditional statistics. In particular, entropy analysis and fractal analysis are non-linear approaches and provide researchers with new opportunities to extract and explore the knowledge hidden in the data. Entropy and fractal dimension are two diametrically opposed but complementary concepts.

Entropy is a measure of chaos. The value of entropy gives an idea of how far the studied object (pattern) is from an ordered, structured state and how close it is to a completely chaotic, structureless, homogeneous form.

Fractal dimension is a metric for characterizing a fractal pattern (which is often a highly organized structure) by quantifying its complexity as the ratio of change in detail to change in scale.

In this study, we use five entropy characteristics and three fractal dimensions, as well as two Hjorth parameters, such as mobility and complexity, to evaluate the patterns of a dataset on ODs in order to identify such indicators that will allow for improving the separation of data from different classes from each other. The identified best indicators can later claim the role of a tool for generating new features used to expand the original dataset. The indicators considered in the proposed study are listed below.

Permutation entropy (PE) [40] is a tool that provides a quantification measure of the complexity of the studied object by capturing the order relations between the values and extracting a probability distribution of the ordinal elements.

Spectral entropy (SPE) [41] is a tool that is based on Shannon's entropy. It measures the irregularity or complexity of the studied object in the frequency domain. After performing a Fourier transform, the studied object is transformed into a power spectrum, and its information entropy presents the power spectral entropy of the studied object.

Singular value decomposition entropy (SVDE) [42] is a tool that characterizes information content or regularity of the studied object depending on the number of vectors attributed to the process.

Approximate entropy (AE) [43–45] is a tool used to quantify the amount of regularity and the unpredictability of fluctuations of the studied object.

Sample entropy (SE) [45] is a modification of approximate entropy. It is used for assessing the complexity of the studied object.

Hjorth mobility (HM) [46,47] is a tool that describes the average frequency for the studied object and provides information about its so-called speed.

Hjorth complexity (HC) [46,47] is a tool that describes the variability of the studied object and refers to the similarity of the studied object to a sinusoidal wave.

Petrosian fractal dimension (PFD) [49] is a tool that allows for estimating the fractal dimension of a finite sequence describing the studied object by means of converting the data to a binary sequence before estimating.

Katz fractal dimension (KFD) [50,51] is a tool which makes exponential transformation of fractal dimension values of the studied object with relative insensitivity to noise.

Higuchi fractal dimension (HFD) [52] is a tool which uses an algorithmic approximate value for the box-counting dimension of the graph of a real-valued function for the studied object.

It is necessary to say that HFD yields a more accurate estimation of the fractal dimension values of the studied object than KFD when tested on synthetic data, but it is more sensitive to noise.

The list of indicators similar to those considered above can be expanded.

Typically, these indicators are used to analyze signals, for example, represented by time series. However, there are also works in which these indicators are used to generate new features of objects based on already known features. For example, the work [10] explores the possibility of using entropies AE and SE to generate new features in the problem of diagnosing ODs based on blood protein markers. Obviously, it is possible to use other indicators in order to extract new features of objects based on already known ones using them.

A more detailed description of the indicators chosen during the experiments will be given in Section 4.

3. A Novel Approach to the Generation of Datasets and the Development of Classifiers

The algorithm for developing the best classifier can be described by the following sequence of steps.

Step 1. Scale each feature of the original dataset to the range [0, 1].

Step 2. Check the features of the dataset for correlation and remove features with high correlation (taking into account their correlation with the target feature that determines class labels for patterns).

Step 3. Calculate the values of potential new features based on the not-scaled dataset, from which the correlated features found in Step 1 are excluded, using five formulas for entropy, two formulas for Hjorth parameters and three formulas for fractal dimension, and choose those that are the best at separating patterns of different classes (based on the average values of entropy and fractal dimensions and average standard deviations). Check the potential new features for correlation and remove features with high correlation (taking into account their correlation with the target feature that determines class labels for patterns).

Step 4. Scale each new feature to the range [0, 1].

Step 5. Set the range [2, H], which will be used in the cycle (during the steps 6–10), where each number h from the range [2, H] is the dimension in the low-dimensional space (the number of features); $2 \le H \le q - 1$; q is the dimension of the original space (the number of features in the original dataset C1).

Step 6. Implement the UMAP algorithm with the number *h* from the range [2, H]. Scale each feature to the range [0, 1]

Step 7. Construct 12 datasets (if h = 2) or 8 datasets (if $h \ge 3$):

1. C1 is the original dataset (with 39 features) (formed only for h = 2).

2. C2 is a dataset based on the UMAP algorithm (from 2 to H features as a result of embedding in a space of lower dimension).

3. C3 is a dataset based on the original dataset and the UMAP algorithm (from 2 to H features).

4. C4 is a dataset based on the UMAP algorithm (from 2 to H features) and one entropy.

5. C5 is a dataset based on the UMAP algorithm (from 2 to H features) and two fractal dimensions.

6. C6 is a dataset based on the UMAP algorithm (from 2 to H features), one entropy and two fractal dimensions.

7. C7 is a dataset based on the original dataset, the UMAP algorithm (from 2 to H features) and one entropy.

8. C8 is a dataset based on the original dataset and one entropy (formed only for h = 2).

9. C9 is a dataset based on the original dataset and two fractal dimensions (formed only for h = 2).

10. C10 is a dataset based on the original dataset, one entropy and two fractal dimensions (formed only for h = 2).

11. C11 is a dataset based on the original dataset, the UMAP algorithm (from 2 to H features) and two fractal dimensions.

12. C12 is a dataset based on the original dataset, the UMAP algorithm (from 2 to H features), one entropy and two fractal dimensions.

Step 8. Rebalance classes for all datasets.

Step 9. Develop classifiers based on the kNN and SVM algorithms. Assess classification quality using cross-validation based on $MacroF_1$ –score. Select the best classifiers based on the kNN and SVM algorithms.

Step 10. Increase the number *h* of dimensions in the low-dimensional space by 1.

Step 11. If $h \le H$, go back to step 6. If h > H, go to step 12.

Step 12. Select the best classifiers based on the results of the algorithm implementation. Complete implementation of the algorithm.

The number of datasets that were used to develop the classifiers was determined as follows.

A total of 12 datasets were used for h = 2. Dataset C1 is the same as the original dataset. Dataset C2 contains features based on the UMAP algorithm. Other datasets were acquired by:

- Adding all possible combinations of three groups of features based on the UMAP algorithm, one entropy and two fractal dimensions (as 1 of 3, 2 of 3, 3 of 3) to the features of the original dataset;
- Adding all possible combinations of two feature groups based on one entropy and two fractal dimensions (as 1 of 2, 2 of 2) to the features based on the UMAP algorithm.

Eight datasets were used for $h \ge 3$, because datasets C1, C8, C9 and C10 do not contain features based on the UMAP algorithm. Namely, the number of generated features depends on *h* whether the UMAP algorithm is used. Thus, it is sufficient to generate the sets C1, C8, C9 and C10 once for h = 2 because their composition does not depend on *h*.

The generation of datasets based on only one entropy and/or two fractal dimensions was not performed due to a significant reduction (convolution) of the initial information in this case.

The choice of the scaling method that implements the transformation of each feature into the range [0, 1] is due to working with the UMAP algorithm, which essentially searches for the coordinates of patterns in a low-dimensional space. In this regard, we decided to scale the values for each coordinate of the patterns in the range [0, 1].

This algorithm uses only one entropy (AE) and only two fractal dimensions (KFD and HFD) because the expediency of using only these out of the 10 indicators specified in Section 2.5 was confirmed experimentally when working with the original dataset based on blood protein markers.

Figure 1 shows a diagram that represents the process of generating the datasets used in the development of classifiers. The upper part of the figure lists the datasets that are generated only once because they are formed without involving the UMAP algorithm.





Figure 2 shows the enlarged block diagram of the classifier development.



Figure 2. The enlarged block diagram of the classifier development.

4. Experimental Studies

During the experiments, the dataset containing information on 39 serum protein markers for 1817 patterns was used. This dataset includes such classes of ODs as breast,

colorectum, esophagus, liver, lung, ovary, pancreas and stomach, as well as the normal class (Norm) corresponding to cases where an OD was not diagnosed. Each protein marker is associated with some feature in the dataset. This dataset is multiclass because it contains information on patterns from nine classes. Accordingly, the classification problem is multiclass and involves the development of a multiclass classifier. This dataset was taken from COSMIC repository [39].

The list of serum protein markers is as follows: AFP (pg/mL), Angiopoietin-2 (pg/mL), AXL (pg/mL), CA-125 (U/mL), CA 15-3 (U/mL), CA19-9 (U/mL), CD44 (ng/mL), CEA (pg/mL), CYFRA 21-1 (pg/mL), DKK1 (ng/mL), Endoglin (pg/mL), FGF2 (pg/mL), Follistatin (pg/mL), Galectin-3 (ng/mL), G-CSF (pg/mL), GDF15 (ng/mL), HE4 (pg/mL), HGF (pg/mL), IL-6 (pg/mL), IL-8 (pg/mL), Kallikrein-6 (pg/mL), Leptin (pg/mL), Mesothelin (ng/mL), Midkine (pg/mL), Myeloperoxidase (ng/mL), NSE (ng/mL), OPG (ng/mL), OPN (pg/mL), PAR (pg/mL), Prolactin (pg/mL), sEGFR (pg/mL), sFas (pg/mL), SHBG (nM), sHER2/sEGFR2/sErbB2 (pg/mL), sPECAM-1 (pg/mL), TGFa (pg/mL), Thrombospondin-2 (pg/mL), TIMP-1 (pg/mL) and TIMP-2 (pg/mL).

All the experiments were conducted using software written in Python 3.10 in the interactive cloud environment Google Colab. The choice of the Python 3.10 programming language can be justified by a large number of various available libraries, including libraries that implement machine learning algorithms.

4.1. Data Analysis Based on the UMAP Algorithm

A preliminary visual analysis of the dataset was performed using the non-linear dimensionality reduction algorithm UMAP. Information on the software implementation of the UMAP algorithm used is available in [70].

With the help of the UMAP algorithm, a 39-dimensional nine-class dataset was embedded into a two-dimensional space (Figure 3). Analysis of the visualization results indicates a complex organization of nine classes in the dataset and their poor separability in general. However, one can try to find such classes in this nine-class dataset that can be well separated from each other.



Figure 3. Visualization of nine-class dataset of ODs using the UMAP algorithm. (*n_neighbors* = 15, *min_dist* = 0.1, *random_state* = 42, *metric* = '*euclidean*').

For example, if only three classes are left in the dataset, such as normal, liver and ovary, then the results of the visualization of the three-class dataset (Figure 4) allow us to conclude that it is expedient to carry out research on the development of a classifier capable of separating patterns of these three classes with high values of classification quality metrics. It should be noted that a three-class dataset with the same list of classes (normal, liver and ovary) was considered by the authors in [10].



Figure 4. Visualization of three-class dataset of ODs using the UMAP algorithm. (*n_neighbors* = 15, *min_dist* = 0.1, *random_state* = 42, *metric* = '*euclidean*').

At the time when this article was written, the ratio of classes in the dataset was Normal:Liver:Ovary = 812:44:54 (which differed slightly from the ratio in the work [10]: Normal:Liver:Ovary = 799:44:54).

It can be assumed that the insufficient separability of patterns of different classes, both in the nine-class dataset and in sets with a smaller number of classes (including in the three-class dataset), is due to the presence of different ODs stages in the patterns presented in the dataset (for example, in the early stages of an OD, separability may be worse). Clearly, further research is needed to answer this question. Insufficiently good separability of patterns of different classes may be due to the insufficiency of the number of features. This problem can be solved by enriching the dataset with new information, either by involving new features in the analysis, for example, based on gene biomarkers, or by generating new features based on existing data, i.e., by extracting knowledge hidden in existing data. In this study, the second approach to enriching the dataset with new information was implemented: an attempt was made to generate new features based on the UMAP algorithm, entropy characteristics and fractal dimensions of the existing data patterns, followed by the selection of new features that satisfy certain a priori selected criteria.

Scaling to the range [0, 1] for each feature was applied to the nine-class and three-class datasets before visualization.

It should be noted that the UMAP algorithm is usually used to embed multidimensional objects in two- or three-dimensional space for visualization, but the results of its application can also be used in the development of data classifiers. Since the visualization of a reduced three-class dataset on ODs in a two-dimensional space indicates the presence of patterns for which a two-dimensional embedding does not allow for distinguishing some objects in their a priori known classes (Figure 2), then it is advised to study various options for embedding, that is, nesting in spaces, whose dimension is equal to the number h, where $h \in \{2, 3, ..., H\}$, H is the maximum dimension of the embedded space. For example, H can be equal to q - 1, where q is the dimension of the original feature space (in this example q = 39). It is also necessary to develop a data classifier for q = 39.

As a result, a group of classifiers will be obtained, developed on the basis of datasets "embedded" in the space of a smaller dimension with the number of features h ($h \in \{2, 3, ..., H\}$), as well as a dataset located in the original feature space (i.e., for q = 39). It will be possible to choose the best classifier from this group in terms of maximizing the classification quality metric (for example, $MacroF_1$ –*score* in the case of working with class-imbalanced datasets). Moving to a lower dimensional space may potentially make it possible to improve the separability of classes from each other, even without making any additional effort, such as balancing classes or taking the sensitivity to the cost of wrong decisions into account.

The scaling of each feature to the range [0, 1] was performed twice during the implementation of the UMAP algorithm: before applying the UMAP algorithm and during preparations of the dataset obtained using the UMAP algorithm for developing a classifier. The resulting dataset can be used on its own or to form an augmented dataset. We can add new features obtained using the UMAP algorithm or extracted in some other way (e.g., by computing entropy characteristics and fractal dimensions of the original dataset) to the features of the original 39-feature dataset. The data obtained by reducing the dimensionality of the original dataset can be considered as new features.

4.2. Generation of New Features Based on Entropies, Hjorth Parameters and Fractal Dimensions of Data Patterns

The original three-class dataset was examined for feature correlation before developing classifiers based on variously formed datasets. The examination showed the absence of a strong correlation with the values of the correlation index of at least 0.7 between the features. The maximum value of the correlation index, equal to 0.604, was found only for one pair of traits with numbers 34 and 35 (sHER2/sEGFR2/sErbB2 (pg/mL) and sPECAM-1 (pg/mL)). The values of the correlation index for other pairs of features turned out to be less than 0.6. As a result, the expediency of using all the features in the further analysis and development of classifiers was proved.

In order to improve the quality of data classification, it was decided to generate new features based on metrics such as entropy characteristics and fractal dimensions of data patterns, selecting among them those that do not correlate with each other or the features of a three-class dataset.

Three-class not scaled to range [0, 1] for each feature dataset was used to generate new features based on the entropy characteristics, Hjorth parameters and fractal dimensions of data patterns. Formulas that were used in the generation of new features for each pattern involved 10 metrics. There were five formulas for entropy, such as permutation entropy (PE), spectral entropy (SPE), singular value decomposition entropy (SVDE), approximate entropy (AE) and sample entropy (SE); two formulas for Hjorth parameters, such as Hjorth mobility and complexity (HM and HC); and three formulas for calculating such fractal dimensions as Petrosian fractal dimension (PFD), Katz fractal dimension (KFD) and Higuchi fractal dimension (HFD).

The results of the calculations of the entropy values, Hjorth parameters and fractal dimensions were grouped into three classes. For each of the three classes, the mean value and the mean standard deviation were calculated for each potential new generated feature.

Comparative analysis of the mean values and mean standard deviations of the aforementioned metrics for each of the three classes for each potential new generated feature (Tables 1 and 2) made it possible to draw the following conclusions. The largest differences between the classes are shown by the entropies AE and SE (Table 1), as well as the fractal dimensions KFD and HFD (Table 1). These metrics were chosen for further consideration. Meanwhile, the mean standard deviations for all the metrics above turned out to be small (and only for the KFD metric they are slightly larger than for other metrics).

					Mean	Values				
Class			Entropy			Hjorth Parameters		Fractal Dimension		
	PE	SPE	SVDE	AE ¹	SE	HM	HC	PFD	KFD	HFD
Normal	0.977	0.921	0.943	0.454	0.508	1.335	1.334	1.068	1.609	2.337
Liver	0.981	0.913	0.937	0.257	0.425	1.326	1.336	1.069	1.548	2.271
Ovary	0.978	0.924	0.975	0.353	0.348	1.313	1.306	1.065	1.696	2.372

Table 1. Mean values for each of the three classes for each potential new generated feature.

¹ Bold type indicates the mean values of the metrics that make it possible to distinguish between classes.

Table 2. Mean standard deviations for each of the three classes for each potential new generatedfeature.

				Ν	lean Standa	rd Deviatio	ns			
Class			Entropy			Hjorth Pa	arameters	Fra	ctal Dimens	ion
	PE	SPE	SVDE	AE ²	SE	HM	HC	PFD	KFD	HFD
Normal	0.018	0.054	0.043	0.091	0.164	0.131	0.510	0.005	0.174	0.068
Liver	0.013	0.085	0.082	0.105	0.105	0.098	0.021	0.005	0.266	0.138
Ovary	0.015	0.026	0.021	0.097	0.131	0.110	0.207	0.004	0.230	0.091

 2 Bold type indicates the mean standard deviations of the metrics that make it possible to distinguish between classes.

The selected metrics were tested for correlation with each other. The tests showed a correlation between the metrics AE and SE (with the value of the correlation metric equal to 0.931). The metric SE was excluded from further consideration, among other things, because it has a lower correlation with the target feature that determines the labels of pattern classes (the values of the correlation metric for the metrics AE and SE are 0.360 and 0.320, respectively, which corresponds to a moderate direct linear dependence on the Chaddock scale). It should be noted that the experiments confirmed the advantage, albeit insignificant, of the metric AE as a tool for generating the values of a new feature included in the dataset (in terms of ensuring a higher quality of data classification). The correlation between the chosen fractal dimensions KFD and HFD is small: it is only 0.141.

Thus, it is advisable to use one feature based on the approximation entropy AE, as well as two features based on fractal dimensions KFD and HFD.

Below, we briefly describe the algorithms that allow for calculating approximation entropy AE, the Katz fractal dimension KFD and the Higuchi fractal dimension HFD.

The algorithm for determining the approximation entropy AE can be described as follows [45].

Suppose we have a sequence of numbers $u = \{u(1), u(2), ..., u(q)\}$ of length q, a non-negative integer ξ ($\xi \le q$) and a positive real number r.

First, the algorithm defines the blocks $\chi(i) = \{u(i), u(i+1), \dots, u(i+\xi-1)\}$ and $\chi(j) = \{u(j), u(j+1), \dots, u(j+\xi-1)\}$, and calculates the distance between $\chi(i)$ and $\chi(j)$ as $d(\chi(i), \chi(j)) = \max_{\kappa = \overline{1,\xi}} (|u(i+\kappa-1) - u(j+\kappa-1)|)$. Then it calculates the value

 $C_i^{\xi}(r) = \frac{number(d(\chi(i),\chi(j)) \le r)}{q - \xi + 1}$, where $j \le q - \xi + 1$. The numerator of $C_i^{\xi}(r)$ defines the

number of blocks of consecutive values of length ξ , which are similar to a given block. As a result, the algorithm calculates the value $\phi^{\xi}(r)$ as

$$\phi^{\xi}(r) = \frac{1}{q-\xi+1} \cdot \sum_{i=1}^{q-\xi+1} \log C_i^{\xi}(r)$$
(16)

and approximation entropy $AE(q, \xi, r)(u)$ as

$$AE(q,\xi,r)(u) = \phi^{\xi}(r) - \phi^{\xi+1}(r), \qquad (17)$$

where $\xi \ge 1$; AE(*q*, 0, *r*)(*u*) = $-\phi^1(r)$.

Approximation entropy $AE(q, \xi, r)(u)$ defines the logarithmic frequency with which blocks of length ξ that are close together stay together for the next position.

 $AE(q, \xi, r)$ is the statistical assessment of the parameter $AE(\xi, r)$:

$$\operatorname{AE}(\xi, r)(u) = \lim_{q \to \infty} \left[\phi^{\xi}(r) - \phi^{\xi+1}(r) \right].$$
(18)

In the proposed research, we used $\xi = 2$ and r = 0.25 (these are values that are usually applied).

In the context of the problem under consideration, we use the description of a certain pattern $x_i \in X$ ($i = \overline{1, s}$; s is the number of patterns in the dataset X) based on blood protein markers corresponding to q features as a certain sequence of numbers u of length q.

The algorithm for determining the Katz fractal dimension KFD can be described as follows [51].

Suppose we have a sequence of points (ζ_i, ϑ_i) of length *q*.

First, the algorithm defines the length *L* of the waveform as

$$L = \sum_{j=0}^{q-2} \sqrt{(\vartheta_{j+1} - \vartheta_j)^2 + (\zeta_{j+1} - \zeta_j)^2}$$
(19)

and the maximum distance Δ between the initial point (ζ_1, ϑ_1) to the other points as

$$\Delta = \max_{j=\overline{2,q}} \sqrt{\left(\zeta_j - \zeta_1\right)^2 - \left(\vartheta_j - \vartheta_1\right)^2}.$$
(20)

Then the algorithm calculates the Katz fractal dimension KFD as

$$D = \frac{\log(q)}{\log(q) + \log\left(\frac{\Delta}{L}\right)}.$$
(21)

In the context of the problem being solved, we consider a sequence of points (j, x_i^j) , where j is the j-th number of the feature in the dataset X ($j = \overline{1,q}$; q is the number of features), as a sequence of points (ζ_j, ϑ_j) of length q; and x_i^j is the value of the j-th feature of the i-th pattern $x_i \in X$ ($i = \overline{1,s}$; s is the number of patterns in the dataset X based on blood protein markers;).

The algorithm for determining the Higuchi fractal dimension HFD can be described as follows [52].

Suppose we have a sequence of numbers $u = \{u(1), u(2), ..., u(q)\}$ of length q.

First, the algorithm defines new sequences u_{κ}^{ς} , defined as:

$$u_{\kappa}^{\xi}; u(\xi), \ u(\xi+\kappa), u(\xi+2\kappa), \dots, u\left(\xi + \left[\frac{q-\xi}{\kappa}\right]\kappa\right) \ (\xi = \overline{1,\kappa}), \tag{22}$$

where [o] is the Gauss' notation, which denotes the integer part of o; ξ is the integer defining the initial moment; and κ is the integer defining the interval moment.

As a result, the algorithm defines κ sets of new sequences. Then the algorithm calculates the length $L_{\xi}(\kappa)$ of curve u_{κ}^{ξ} as

$$L_{\xi}(\kappa) = \frac{q-1}{\left[\frac{q-\xi}{\kappa}\right] \cdot \kappa^2} \cdot \sum_{i=1}^{\left[\frac{q-\xi}{\kappa}\right]} |u(\xi+i\cdot\kappa) - u(\xi+(i-1)\cdot\kappa)|,$$
(23)

and the length $L(\kappa)$ as

$$L(\kappa) = \frac{1}{\kappa} \cdot \sum_{\xi=1}^{\kappa} L_{\xi}(\kappa).$$
(24)

Then the algorithm calculates the Higuchi fractal dimension HFD as the slope of the best-fitting linear function through the data points:

$$\left\{ \left(\log \frac{1}{\kappa}, \log(L(\kappa)) \right) \right\}.$$
 (25)

In the context of the problem under consideration, we use the description of a certain pattern $x_i \in X$ ($i = \overline{1, s}$; s is the number of patterns in the dataset X) based on blood protein markers, corresponding to q features, as a certain sequence of numbers u of length q.

In the proposed research, we used values for $\kappa \leq 10$ and m < k.

4.3. Generation of Datasets Used in the Development of Classifiers

The development of the classifiers was carried out based on the following datasets generated based on new features from Sections 4.1 and 4.2:

- 1. C1 is the original dataset (it contains 39 features);
- 2. C2 is a dataset based on the UMAP algorithm (it contains from 2 to *H* features as a result of embedding in a space of lower dimension);
- 3. C3 is a dataset based on the original dataset and the UMAP algorithm (it generates from 2 to *H* features);
- 4. C4 is a dataset based on the UMAP algorithm (it generates from 2 to *H* features) and one entropy;
- 5. C5 is a dataset based on the UMAP algorithm (it generates from 2 to *H* features) and two fractal dimensions;
- 6. C6 is a dataset based on the UMAP algorithm (it generates from 2 to *H* features), one entropy and two fractal dimensions;
- 7. C7 is a dataset based on the original dataset, the UMAP algorithm (it generates from 2 to *H* features) and one entropy;
- 8. C8 is a dataset based on the original dataset and one entropy;
- 9. C9 is a dataset based on the original dataset and two fractal dimensions;
- 10. C10 is a dataset based on the original dataset, one entropy and two fractal dimensions;
- 11. C11 is a dataset based on the original dataset, the UMAP algorithm (it generates from 2 to *H* features) and two fractal dimensions;
- 12. C12 is a dataset based on the original dataset, the UMAP algorithm (it generates from 2 to *H* features), one entropy and two fractal dimensions.

The content of the datasets (namely, the number and selection of features) C1, C8, C9 and C10 does not depend on the dimension h of the space into which the 39-dimensional feature space of the original dataset is embedded when applying the UMAP algorithm. Therefore, classifiers based on these datasets should be developed once. Balancing algorithms, such as SMOTE and its modifications that implement the synthesis of new patterns at the classes' boundary (Borderline SMOTE-1, Borderline SMOTE-2 and ADASYN), are applied once, as well. After this is completed, new classifiers are developed.

The number and selection of features in the remaining datasets C2, C3, C4, C5, C6, C7, C11 and C12 depends on the dimension h of the space into which the UMAP algorithm embeds the 39-dimensional feature space of the original dataset. Therefore, new classifiers

should be developed based on the datasets C2, C3, C4, C5, C6, C7, C11 and C12 for each *h*, both in the case of refusal to use class balancing algorithms, and in the case of their application.

If the dataset is supposed to use a feature based on the entropies, then two variants of the classifier are developed in order to assess the advantages of using the approximation entropy AE and the sample entropy SE in relation to each other, followed by choosing the best entropy for the role of the entropy used for generation of new feature values.

If the UMAP algorithm is not used in the formation of the dataset, i.e., the dataset does not depend on the dimension h of the space into which the 39-dimensional feature space of the original dataset is embedded, then we will identify the names of the classifiers with the names of the datasets corresponding to them. In this case, we will discuss classifiers C1, C8, C9 and C10. If the UMAP algorithm is used when forming a dataset, i.e., the dataset depends on the dimension of the space h into which the 39-dimensional feature space of the original dataset is embedded using the UMAP algorithm, then we will add an indication of the space dimension to the name of the corresponding dataset. For example, we will talk about the C3 classifier (for h = 5), if, during its development, a dataset was used that was formed based on the results of applying the UMAP algorithm for h = 5.

4.4. Aspects of k-Fold Cross-Validation

The classifiers were developed using the kNN and SVM algorithms, the software implementations of which were taken from the scikit-learn library of the Python language.

It should be noted that it is possible to use other machine learning algorithms, for example, the RF algorithm, but this can lead to significant time costs for the development of classifiers due to the specifics of the algorithm itself.

First of all, we developed classifiers for different values of the dimension *h* of the space in which the UMAP algorithm embedded the original 39-dimensional space. The classifiers were developed using the kNN and SVM algorithms based on 12 datasets. No balancing algorithms had been applied to the datasets prior to that.

Then we developed classifiers using datasets that were balanced by classes based on four algorithms: SMOTE, Borderline SMOTE-1, Borderline SMOTE-2 and ADASYN. In this case, we used the kNN and SVM algorithms once again.

Before balancing, the ratio of classes in each of the 12 datasets was Normal:Ovary:Liver = 812:54:44.

The classes were balanced with the values of the parameters of the balancing algorithms set by default in the Python program libraries.

After balancing the classes using the SMOTE algorithm, the ratio of classes in each of the 12 datasets became Normal:Ovary:Liver = 812:812:812.

After balancing the classes using the Borderline SMOTE-1 algorithm, the ratio of classes in each of the 12 datasets became Normal:Ovary:Liver = 812:812:812.

After balancing the classes using the Borderline SMOTE-2 algorithm, the ratio of classes in each of the 12 datasets became Normal:Ovary:Liver = 812:812:811.

After balancing the classes using the ADASYN algorithm, the ratio of classes in each of the 12 datasets became Normal:Ovary:Liver = 812:807:806.

In order to assess the quality of each classifier, the k-fold cross-validation procedure, which is an effective approach for estimating the performance of a classifier, was applied.

 $MacroF_1$ -score was used as the main metric of classification quality in order to reduce the negative impact on the quality of classification of the existing class imbalance in the original C1 dataset.

A grid search was implemented for the optimal values of the parameters of the kNN and SVM classifiers using the classical approach to the implementation of the k-fold cross-validation procedure. In this case, the k classifiers are trained and evaluated on the k holdout test sets. As a result, the mean performance of the k classifiers is evaluated.

We proposed to perform 10-fold cross-validation (that is k = 10) during implementation of the grid search for the optimal values of the parameters of the best classifier. We used a stratified sampling strategy [71,72]. The results of the cross-validation were used to calculate the mean value of $MacroF_1$ -score and the corresponding standard deviation. For the best classifier, similar values were calculated for such metrics as *Accuracy*, *MacroPrecision* and *MacroRecall*. In addition, hyperparameter values were determined for the best classifier.

A grid search was implemented while working with the kNN algorithm. The following parameters were used in the grid search: $n_neighbors$, which corresponds to the number of nearest neighbors, and *weights*, which corresponds to weight coefficients assigned to the neighbors. The value of the number of neighbors $n_neighbors$ varied from 5 to 15 with a step of one. The *weights* parameter could take one of two values: '*uniform*' and '*distance*'. In the first case, all neighbors of some object had equal weights. In the second case, the neighbor's weight depended on the distance to the object: the smaller the distance, the greater the weight. As for the rest of the parameters, we used the default values set in the software implementation of the kNN algorithm in the Python scikit-learn library. As a result, 10 * (11 * 2) = 220 model evaluations were obtained with a single pass through the grid.

Working with the SVM algorithm involved the implementation of a grid search, as well. In this case, it was implemented for the values of such parameters as: *gamma*, which is a parameter of the radial basis function of the kernel, and *C*, which is a regularization parameter. The value of the *gamma* parameter varied from 0.4 to 2 with a step of 0.1. The value of parameter *C* also changed from 0.4 to 2 with a step of 0.1.

We used the default values set in the software implementation of the SVM algorithm in the Python scikit-learn library as the values of the rest parameters. As a result, 10 * (17 * 17) = 2890 model evaluations were obtained with a single pass through the grid.

4.5. Development of the Classifiers

We conducted research in order to determine the feasibility of using the approximation entropy or sample entropy when forming the values of new features for each of the kNN and SVM algorithms used in the development of the classifiers. The feasibility assessment was performed for both datasets that were not subjected to class balancing, and for datasets that were subjected to class balancing using four algorithms: SMOTE, Borderline SMOTE-1, Borderline SMOTE-2 and ADASYN.

The preference for one or another entropy was given based on its provision of the maximum mean value of $MacroF_1$ -score at the test sets on a group of the kNN or SVM classifiers developed on the basis of the studied datasets (without class balancing or with balancing using one of the four algorithms). The best class balancing algorithm was chosen for each of the kNN and SVM algorithms used in the development of the classifiers.

The results of our research for the kNN and SVM algorithms used in the development of the classifiers are shown in Tables 3 and 4, respectively. The development of the classifiers was carried out for h = 2, ..., 38. The maximum mean values of $MacroF_1$ -score in columns AE and SE for each type of classifier are highlighted in bold, and the coinciding values are italicized.

Table 3. Stuc	ly of the a	dvantages of	entropies	AE and SE in	the development	of the kNN classifiers
	2	0			1	

Type of Classification Algorithm/	Maximum Mean Value of MacroF ₁ -score				
Class Balancing Algorithm	AE	SE			
kNN/no class balancing	0.842	0.849 ³			
kNN/SMOTE	0.866	0.864			
kNN/Borderline SMOTE-1	0.878	0.878			
kNN/Borderline SMOTE-2	0.861	0.861			
kNN/ADASYN	0.842	0.842			

³ The metric value with the largest value in the row is highlighted in bold. Matching metric values in a row are italicized.

Type of Classification Algorithm/	Maximum Mean Value of MacroF ₁ -score			
Class Balancing Algorithm	AE	SE		
SVM/no class balancing	0.883	0.884 ⁴		
SVM/SMOTE	0.912	0.912		
SVM/Borderline SMOTE-1	0.911	0.911		
SVM/Borderline SMOTE-2	0.886	0.886		
SVM/ADASYN	0.905	0.905		

Table 4. Study of the advantages of entropies AE and SE in the development of the SVM classifiers.

⁴ The metric value with the largest value in the row is highlighted in bold. Matching metric values in a row are italicized.

The experimental results did not reveal a clear advantage of the approximation entropy over the sample entropy. Preference was given to the approximation entropy because of its higher correlation with the target feature. However, it is possible that, in the case of working with the SVM algorithm, preference should be given to the sample entropy SE due to the fact that the time spent on calculating the sample entropy SE is less than calculating the approximate entropy AE.

Based on the results of the analysis of Tables 3 and 4, a class balancing algorithm was also identified, which allowed for obtaining larger maximum mean values of the $MacroF_1$ -score. This is the Borderline SMOTE-1 algorithm for the kNN classifier development (Table 3), and the SMOTE algorithm for the SVM classifier development (Table 4). These algorithms will be considered in subsequent detailed studies when developing the corresponding classifiers.

4.6. Development of kNN Classifiers

Euclidean distance metric was used during development of the kNN classifiers. The weights of neighbors for each analyzed object could be equal or dependent on the distance to this object. The Borderline SMOTE-1 algorithm was used to implement class balancing.

4.6.1. Experiment without Class Balancing

Figure 5 presents the results of the experiment in choosing the best kNN classifier in the case of working with the approximation entropy AE when forming some of the datasets used in the development of the classifiers. The balancing of classes in datasets was not performed here.



Figure 5. Visualization of the results of the experiment of choosing the best kNN classifier based on 12 datasets without using a class balancing algorithm ($n_neighbors$ is the number of nearest neighbors; *weights* is the parameter which assigns weight coefficients to the neighbors; *q* is the dimension of the space corresponding to the dataset used for development of classifier; *h* is the dimension of the space into which the UMAP algorithm embeds the 39-dimensional feature space corresponding to the original dataset; the background of each color shows the amount of standard deviation around the mean of the metric $MacroF_1$ –score).

The red color indicates the line corresponding to the mean value of the $MacroF_1$ -score obtained for the best C1 classifier developed on the basis of the original dataset, i.e., the dataset with 39 features. The light red shading shows the spread for the $MacroF_1$ -score mean value calculated from its standard deviation. The blue color indicates the line corresponding to the mean values of the $MacroF_1$ -score obtained for the best classifiers developed on the basis of the modified datasets. The light blue shading shows the spread for the $MacroF_1$ -score means calculated from their standard deviation. In addition, the following information is presented in Figure 5 next to the names of the classifiers developed

on the basis of datasets: the number of features that depend on the dimension h of the space into which the original 39-dimensional space is embedded, the dimensions h of the space allowing for building the best classifiers, the final dimension q of the space corresponding to the dataset used for development of classifier, and the best values of classifiers parameters are indicated. The same designations are used in Figures 6–8.



Figure 6. Visualization of the results of the experiment of choosing the best kNN classifier based on 12 datasets using the Borderline SMOTE-1 class balancing algorithm (*n_neighbors* is the number of nearest neighbors; *weights* is the parameter which assigns weight coefficients to the neighbors; *q* is the dimension of the space corresponding to the dataset used for the development of the classifier; *h* is the dimension of the space into which the UMAP algorithm embeds the 39-dimensional feature space corresponding to the original dataset; the background of each color shows the amount of standard deviation around the mean of the metric $MacroF_1-score$).



Figure 7. Visualization of the results of the experiment for choosing the best SVM classifier based on 12 datasets without using class balancing algorithms (*gamma* is the parameter of the radial basic kernel function; *C* is the regularization parameter; *q* is the dimension of the space corresponding to the dataset used for development of the classifier; *h* is the dimension of the space into which the UMAP algorithm embeds the 39-dimensional feature space corresponding to the original dataset; the background of each color shows the amount of standard deviation around the mean of the metric $MacroF_1$ –score).

1.00

0.95

0.90

0.85

0.80

0.75

0.70

0.65

MacroF1 - score





Figure 8. Visualization of the results of the experiment for choosing the best SVM classifier based on 12 datasets using the SMOTE algorithm for class balancing (*gamma* is the parameter of the radial basic kernel function; *C* is the regularization parameter; *q* is the dimension of the space corresponding to the dataset used for the development of the classifier; *h* is the dimension of the space into which the UMAP algorithm embeds the 39-dimensional feature space corresponding to the original dataset; the background of each color shows the amount of standard deviation around the mean of the metric *MacroF*₁–*score*).

As can be seen from Figure 5, all the classifiers developed on the basis of the modified datasets outperformed C1 classifier developed on the basis of the original dataset in terms of the mean value of the $MacroF_1$ –score.

Classifier C6 (with h = 24) turned out to be the best: it has a mean value of *MacroF*₁-*score* equal to 0.842 (with a standard deviation of 0.080), while classifier C1 has mean value of *MacroF*₁-*score* equal to 0.756 (while the standard deviation is 0.101).

The dataset used in the development of classifier C6 (with h = 24) was obtained from the original one as a result of applying the UMAP algorithm to it with h = 24. This dataset contains 27 features in total, including one feature based on approximation entropy AE and two features based on fractal dimensions KFD and HFD.

Classifiers C2 (with h = 36), C4 (with h = 11) and C5 (with h = 19) also turned out to be relatively good in terms of the mean value of the *MacroF*₁-*score*.

The worst classifier in this experiment is classifier C9 (independent of *h*) developed on the basis of the dataset obtained by adding two features based on fractal dimensions KFD and HFD to the original dataset.

Table 5 shows the main characteristics of classifier C1, as well as the best classifier, namely, classifier C6 (with h = 24), without class balancing.

Table 5. Characteristics of kNN classifiers C1 and C6 (with h = 24) in the experiment without class balancing.

	Classifier				
Characteristic —	C1	C6 (with <i>h</i> = 24)			
Number of features in the dataset	39	27			
Number of neighbors (n_neighbors)	6	12			
weights	'distance'	'distance'			
$MacroF_1$ -score (mean/std)	0.756/0.101	0.842/0.093			
Accuracy (mean/std)	0.948/0.017	0.966/0.016			
MacroRecall (mean/std)	0.687/0.106	0.803/0.091			
MacroPrecision (mean/std)	0.938/0.063	0.919/0.098			
Training time (mean/std), s.	0.002/0.001	0.004/0.002			
Quality metrics calculation time (mean/std), s.	0.009/0.003	0.009/0.002			

In the experiment under consideration, the use of the modified dataset C6 (with h = 24) obtained from the original dataset C1 allowed for increasing the value of the *MacroF*₁-*score* for classifier C6 (with h = 24) by 0.086 compared to classifier C1 (the standard deviation for metric *MacroF*₁-*score* of classifier C6 (with h = 24) turned out to be less than that of classifier C1). The training time of classifier C6 (with h = 24) increased about two times. The quality metrics calculation time during the testing did not change much.

It should be noted that classifier C6 (with h = 24), as well as classifiers C2 (with h = 36), C4 (with h = 11) and C5 (with h = 19), outperformed the classifier developed in [10] using the principles of cost-sensitive algorithms based on the mean values of the main quality metrics. At the same time, one can notice slight discrepancies in the number of patterns of the normal class in the proposed study and in [10]: in our dataset there are 13 more such patterns, but this could only negatively affect our results (compared to the results in [10]), which, however, did not happen.

The F_1 of the best classifier in [10] was equal to 0.819, and the values of such metrics as *Accuracy*, *Recall* and *Precision* were equal to 0.952, 0.807 and 0.833, respectively. Unfortunately, the rules for choosing the best classifier in our study and in [10] may be somewhat different (for example, we do not know if standard deviation estimates were calculated in that study), but we assume that our best classifiers (with the best (maximum) values of quality metrics) clearly outperform the best classifier in [10]. To confirm these conclusions,

we will provide additional information on classifiers C2 (with h = 36), C4 (with h = 11) and C5 (with h = 19) (because for classifier C6 (with h = 24), such information is given in Table 5).

Classifier C2 (with h = 36) has a mean value of $MacroF_1$ -score equal to 0.841 (with a standard deviation of 0.093) and mean values of such metrics as *Accuracy*, *Recall* and *Precision* equal to 0.965 (with a standard deviation of 0.015), 0.812 (with a standard deviation of 0.088) and 0.917 (with a standard deviation of 0.065), respectively.

Classifier C4 (with h = 11) has a mean value of $MacroF_1$ -score equal to 0.839 (with a standard deviation of 0.094) and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.963 (with a standard deviation of 0.016), 0.818 (with a standard deviation of 0.094) and 0.900 (with a standard deviation of 0.098), respectively.

Classifier C5 (with h = 19) has a mean value of $MacroF_1$ -score equal to 0.839 (with a standard deviation of 0.083) and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.964 (with a standard deviation of 0.017), 0.800 (with a standard deviation of 0.095) and 0.930 (with a standard deviation of 0.058), respectively.

Thus, we can conclude that even a simple addition of new features can increase the separability of patterns of one class from patterns of another class.

Analysis of the values of the classification quality metrics (Table 5), in particular, the values of the *MacroRecall* metric, allowed us to conclude that the quality of the classification is not high enough and that it is necessary to take additional actions to improve the quality of the classification. In order to perform such actions, we can use class balancing or costsensitive algorithms. In this study, we implemented algorithms for balancing classes in datasets.

4.6.2. Class Balancing Experiment

Figure 6 presents the results of the experiment for choosing the best kNN classifier in the case of working with the approximation entropy AE in the formation of some of the 12 datasets used in the development of the classifiers. In this case, the balancing of classes in datasets was performed using the Borderline SMOTE-1 algorithm.

The red color indicates the line corresponding to the mean value of the $MacroF_1$ -score obtained for the best classifier C1 developed on the basis of the original dataset, i.e., the dataset with 39 features. The light red shading shows the spread for the $MacroF_1$ -score mean value, calculated from its standard deviation. The blue color indicates the line corresponding to the mean values of the $MacroF_1$ -score obtained for the best classifier C1 developed on the basis of the balanced original dataset. The light blue shading shows the spread for the $MacroF_1$ -score means, calculated from their standard deviation. The green color indicates the line corresponding to the mean values of the mean values of $MacroF_1$ -score obtained for the best classifiers developed on the basis of the mean values of $MacroF_1$ -score obtained for the best classifiers developed on the basis of the mean values of the modified balanced datasets. The light green shading shows the spread for the $MacroF_1$ -score means, calculated from their standard deviation.

Figure 6 shows that all the classifiers developed on the basis of the balanced datasets provide a higher mean value of the $MacroF_1$ -score than the classifier C1 developed on the basis of the original dataset. At the same time, some classifiers have lower values of the $MacroF_1$ -score than the classifier C1 developed on the basis of the balanced original dataset. In addition, Figure 6 shows a decrease in the standard deviations for the $MacroF_1$ -score for the classifiers developed on the basis of balanced datasets compared to the standard deviations for the $MacroF_1$ -score for the classifiers developed on the basis of imbalanced datasets (Figure 5).

Classifier 8 (independent of *h*) turned out to be the best: it has a mean value of $MacroF_1$ -score equal to 0.878 (with a standard deviation of 0.050), while classifier C1 has a mean value of $MacroF_1$ -score equal to 0.847 (while the standard deviation is 0.079).

The dataset used in the development of classifier C8 (independent of h) was obtained from the original one by adding one feature based on approximation entropy AE. This dataset contains 40 features in total. The classifiers C7 (with h = 4), C10 (independent of h), C11 (with h = 18) and C12 (with h = 19) also turned out to be relatively good in terms of the mean value of the

*MacroF*₁-*score*. The worst classifiers in this experiment were classifier C4 (with h = 11) and C6 (with h = 12). The first classifier was developed on the basis of a dataset obtained from the original one as a result of applying the UMAP algorithm to it with h = 11 by adding one feature based on approximation entropy AE. The second classifier was developed on the basis of a dataset obtained from the original one as a result of applying the UMAP algorithm to it with h = 12 by adding one feature based on approximation entropy AE. The second classifier was developed on the basis of a dataset obtained from the original one as a result of applying the UMAP algorithm to it with h = 12 by adding one feature based on approximation entropy AE and two features based on fractal dimensions KFD and HFD.

From Figures 5 and 6, one can notice a decrease in the standard deviation of classifiers in the case of using the Borderline SMOTE-1 algorithm to balance classes in datasets.

Table 6 shows the main characteristics of classifier C1, as well as the best classifier, namely classifier C8 (independent of h), with class balancing.

Classifier Characteristic C1 C8 (independent of h) Number of features in the 39 40 dataset Number of neighbors 10 6 (*n_neighbors*) weights 'uniform' 'uniform' 0.847/0.079 0.878/0.050 $MacroF_1$ -score (mean/std) Accuracy (mean/std) 0.957/0.022 0.968/0.013 0.870/0.079 MacroRecall (mean/std) 0.877/0.066 MacroPrecision (mean/std) 0.846/0.085 0.896/0.063 Training time (mean/std), s. 0.012/0.006 0.028/0.002 Quality metrics calculation 0.024/0.009 0.017/0.003 time (mean/std), s.

Table 6. Characteristics of kNN classifiers C1 and C8 (independent of *h*) in the experiment using the Borderline SMOTE-1 class balancing algorithm.

It should be noted that classifier C8 (independent of h), as well as classifiers C7 (with h = 4), C10 (independent of h), C11 (with h = 18) and C12 (with h = 19) outperformed the classifier developed in [10] using the principles of cost-sensitive algorithms based on the mean values of the main quality metrics.

To confirm these conclusions, we will provide additional information on classifiers C7 (with h = 4), C10 (independent of h), C11 (with h = 18) and C12 (with h = 19) (because for classifier C8 (independent of h), such information is given in Table 6).

Classifier C7 (with h = 4) has mean value of $MacroF_1$ -score equal to 0.871 (with a standard deviation of 0.065), and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.969 (with a standard deviation of 0.014), 0.864 (with a standard deviation of 0.077) and 0.902 (with a standard deviation of 0.067), respectively.

Classifier C10 (independent of *h*) has mean value of $MacroF_1$ –*score* equal to 0.867 (with a standard deviation of 0.070), and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.966 (with a standard deviation of 0.016), 0.870 (with a standard deviation of 0.081) and 0.888 (with a standard deviation of 0.074), respectively.

Classifier C11 (with h = 18) has mean value of $MacroF_1$ -score equal to 0.864 (with a standard deviation of 0.057), and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.966 (with a standard deviation of 0.015), 0.866 (with a standard deviation of 0.064) and 0.883 (with a standard deviation of 0.067), respectively.

Classifier C12 (with h = 19) has mean value of $MacroF_1$ -score equal to 0.865 (with a standard deviation of 0.066), and mean values of such metrics as *Accuracy*, *Recall* and *Precision* are equal to 0.968 (with a standard deviation of 0.014), 0.864 (with a standard deviation of 0.074) and 0.890 (with a standard deviation of 0.064), respectively.

In the experiment under consideration, using the best classifier C8 (independent of *h*) made it possible to increase the mean value of $MacroF_1$ -score by 0.031 compared to the C1 classifier (with the standard deviation for $MacroF_1$ -score of classifier C8 (independent of *h*) being less than that of classifier C1). The training time of classifier C8 increased about 2.3 times. The quality metrics calculation time during the testing even decreased slightly.

Analysis of the values of classification quality metrics (Table 6), particularly the values of the *MacroRecall* metric, allows us to conclude that the classification quality by important metrics has increased. It should be noted that the training time in the case of applying class balancing to the dataset has increased, for example, about two times for classifier C1.

4.7. Development of SVM Classifiers

We used the radial basis kernel function during the development of the SVM classifiers. Experiments with the linear function of the kernel were also carried out, but turned out to be less successful, therefore they are not presented in this study. The SMOTE algorithm was used to implement class balancing.

4.7.1. Experiment without Class Balancing

Figure 7 presents the results of the experiment for choosing the best SVM classifier in the case of working with the approximation entropy AE when forming some of 12 datasets used in the development of the classifiers. The balancing of classes in the datasets was not performed here. Figure 7 uses the same notations as Figure 5.

As can be seen from Figure 7, only classifier C8 (independent of *h*), classifier C9 (independent of *h*) and classifier C10 (independent of *h*) developed on the basis of the modified datasets were able to outperform classifier C1 (which has mean value of $MacroF_1$ –*score* equal to 0.877 with the standard deviation equal to 0.078), developed on the basis of the original dataset by the mean value of the $MacroF_1$ –*score*. At the same time, classifier C8 (independent of *h*) outperformed classifier C9 (independent of *h*) and classifier C10 (independent of *h*), if we compare them by the mean values of the $MacroF_1$ –*score* equal to 0.885 (with the standard deviation equal to 0.079); classifier C9 (independent of *h*) has a mean value of $MacroF_1$ –*score* equal to 0.885 (with the standard deviation equal to 0.079); classifier C9 (independent of *h*) has a mean value of $MacroF_1$ –*score* equal to 0.878 (with the standard deviation of 0.078); and classifier C10 (independent of *h*) has a mean $MacroF_1$ –*score* of 0.880 (with a standard deviation of 0.074). The rest of the classifiers turned out to be less successful than classifier C1.

Note that classifier C8 (independent of *h*), classifier C9 (independent of *h*) and classifier C10 (independent of *h*) are developed on the basis of datasets containing 40 (39 + 1), 41 (39 + 2) and 42 (39 + 1 + 2) features, respectively.

It is obvious that preference should be given to the classifier developed using the dataset with fewer features: this is classifier C8 (independent of *h*). In addition, it has the highest mean value of $MacroF_1$ –score.

The dataset used in the development of classifier C8 (independent of h) was obtained from the original one by adding a new feature formed on the basis of the approximation entropy AE.

Table 7 shows the main characteristics of classifier C1, as well as the best classifier, namely classifier C8, without class balancing.

	Classifier					
Characteristic	C1	C8 (independent of h)				
Number of features in the dataset	39	40				
gamma	1.2	1.2				
С	2.0	2.0				
$MacroF_1$ -score (mean/std)	0.877/0.078	0.885/0.079				
Accuracy (mean/std)	0.973/0.015	0.974/0.016				
MacroRecall (mean/std)	0.843/0.088	0.850/0.090				
MacroPrecision (mean/std)	0.950/0.053	0.957/0.051				
Training time (mean/std), s.	0.123/0.008	0.131/0.007				
Quality metrics calculation time (mean/std), s.	0.007/0.001	0.009/0.001				

Table 7. Characteristics of SVM classifiers C1 and C8 in the experiment without class balancing.

In the experiment under consideration, the use of the best classifier C8 (independent of *h*) made it possible to increase the mean value of the *MacroF*₁-*score* by 0.008 compared to the C1 classifier (with approximately the same standard deviations). The training time of classifier C8 increased slightly, which was expected (because the number of features increased by only 1). The same can be said about the quality metrics calculation time during the testing.

Analysis of the values of the classification quality metrics (Table 7), particularly the values of the *MacroRecall* metric, allows us to conclude that the quality of the classification is not high enough and that it is necessary to take additional action to improve the quality of the classification.

4.7.2. Class Balancing Experiment

Figure 8 presents the results of the experiment for choosing the best SVM classifier in the case of working with the approximation entropy AE when forming some of the 12 datasets used in the development of classifiers. In this case, the balancing of classes in datasets was performed using the SMOTE algorithm.

Figure 8 uses the same notations as Figure 6.

As can be seen from Figure 8, only classifier C3 (with h = 2), classifier C7 (with h =28) and classifier C8 (independent of *h*) developed on the basis of the modified datasets were able to outperform classifier C1 (which has mean value of $MacroF_1$ -score equal to 0.910 with the standard deviation equal to 0.064) developed on the basis of the original dataset by the mean value of the $MacroF_1$ -score. At the same time, classifier C7 (with h = 28) outperformed classifier C3 (with h = 2) and classifier C8 (independent of h), if we compare them by the mean values of $MacroF_1$ -score: for example, classifier C7 (with h = 28) has a mean value of *MacroF*₁*-score* equal to 0.914 (with the standard deviation equal to 0.050); classifier C3 (with h = 2) has a mean value of MacroF₁-score equal to 0.912 (while the standard deviation is 0.058); and classifier C8 (independent of h) has a mean $MacroF_1$ -score of 0.913 (with a standard deviation of 0.061). It should be noted that classifiers C9 (independent of h), C10 (independent of h), C11 (with h = 3) and C12 (with h = 2) outperformed classifier C1 developed on the basis of the original dataset that was not subjected to class balancing (Section 4.7.1) by the mean value of $MacroF_1$ -score. The rest of the classifiers turned out to be inefficient compared to classifier C1 developed on the basis of the original dataset that was not subjected to class balancing (Section 4.7.1) and classifier C1 developed on the basis of the original dataset that was subjected to class balancing (Section 4.7.2).

Note that classifier C3 (with h = 2), classifier C7 (with h = 28) and classifier C8 (independent of h) were developed on the basis of the datasets containing 41 (39 + 2), 68 (39 + 28 + 1) and 40 (39 + 1) features, respectively.

It is obvious that preference should be given to the classifier which has the highest mean value of $MacroF_1$ -score: this is classifier C7 (with h = 28). However, it was developed using the dataset with the greatest number of features among the three datasets discussed above.

The dataset used in the development of classifier C7 (with h = 28) was obtained from the original one (with 39 features) by adding one new feature formed on the basis of the approximation entropy AE and 28 features obtained using the UMAP algorithm.

Alternatively, we can use classifier C3 (with h = 2) and classifier C8 (independent of h), which are less accurate by the mean value of $MacroF_1$ -score, but developed from datasets with fewer features.

It should be noted that the best classifiers, particularly classifier C3 (with h = 2), classifier C7 (with h = 28) and classifier C8 (independent of h), outperformed all the best classifiers proposed in Sections 4.6.1, 4.6.2 and 4.7.1 in terms of the main quality metrics.

From Figures 7 and 8, one can notice a decrease in the standard deviation of classifiers in the case of using the SMOTE algorithm to balance classes in datasets.

Table 8 shows the main characteristics of classifier C1, as well as the best classifier, namely, classifier C7 (with h = 28) with class balancing.

Table 8. Characteristics of SVM classifiers C1 and C7 (with h = 28) in the experiment with class balancing.

-

In the experiment under consideration, the use of the best classifier C7 (with h = 28) made it possible to increase the mean value of the $MacroF_1$ -score by 0.004 compared to the C1 classifier (with approximately the same standard deviations). It should be noted that the training time and quality metrics calculation time during the testing in the case of applying class balancing to the dataset has decreased approximately 1.8 and 1.6 times, respectively, for classifier C7 (with h = 28), despite the increase in the number of features.

Analysis of the values of the classification quality metrics (Table 8), particularly the values of the *MacroRecall* metric, allows us to conclude that the classification quality in terms of the main metrics has increased.

5. Discussion

The results of the experiments with two machine learning algorithms such as kNN and SVM showed the feasibility of modifying the original dataset by adding new features based on the approximation entropy AE and fractal dimensions KFD and HFD, and also

based on the UMAP algorithm, and sometimes by replacing the original dataset with the results of applying the UMAP algorithm to it with the addition of new features based on the approximation entropy AE and fractal dimensions KFD and HFD.

At the same time, due to the high imbalance of classes in the original dataset, it is advisable to use class balancing algorithms and cost-sensitive algorithms. In the proposed study, four class balancing algorithms were implemented (SMOTE, Borderline SMOTE-1, Borderline SMOTE-2 and ADASYN). The Borderline SMOTE-1 algorithm for the kNN classifier and the SMOTE algorithm for the SVM classifier were recognized as the best. However, balancing classes using appropriate algorithms is associated with significant time costs, so the goal of further research is to develop classifiers using cost-sensitive algorithms.

In the context of working with the kNN algorithm using the Borderline SMOTE-1 algorithm for class balancing, classifier C8 (independent of *h*) turned out to be the best.

The dataset used in the development of classifier C8 (independent of *h*) was obtained from the original one by adding one feature based on approximation entropy AE. Thus, the development of the kNN classifier was performed in 40-dimensional space (while the original space was 39-dimensional).

In the context of working with the SVM algorithm using the SMOTE algorithm for class balancing, classifier C7 (with h = 28) turned out to be the best.

The dataset used in the development of classifier C7 (with h = 28) was obtained from the original one by adding new features obtained using the UMAP algorithm to the original dataset with the dimension of the new space h = 28 and one feature based on the approximation entropy AE. Thus, the development of the SVM classifier C7 (with h = 28) was performed in 68-dimensional space (while the original space was 39-dimensional).

All 12 kNN classifiers developed on the basis of class-balanced datasets using the Borderline SMOTE-1 algorithm outperformed the base classifier C1 developed on the basis of the original dataset, in which features were compared to blood protein markers (Figure 4).

Eight out of 12 SVM classifiers developed on the basis of class-balanced datasets using the SMOTE algorithm outperformed the basic C1 classifier developed on the basis of the original dataset, in which features were compared to blood protein markers. These are classifiers C1, C3 (with h = 2), C7 (with h = 28), C8, C9, C10, C11 (with h = 3) and C12 (with h = 2). Four out of 12 SVM classifiers developed on the basis of class-balanced datasets using the SMOTE algorithm turned out to be even worse than the basic C1 classifier developed on the basis of the original dataset. Such classifiers are C2 (with h = 36), C4 (with h = 11), C5 (with h = 36) and C6 (with h = 36) (Figure 6).

The classifiers recognized as the best in Sections 4.6.1 and 4.7.1 outperformed the classifier proposed in [10] in terms of the main quality metrics. However, it was decided to use balancing algorithms in order to restore the balance of classes. The classifiers recognized as the best in Sections 4.6.2 and 4.7.2 outperformed the classifier proposed in [10], as well as the classifiers developed in Sections 4.6.1 and 4.7.1, in terms of the main quality metrics. In general, it should be noted that the proposed approach to the formation of datasets by generating new features using different tools with their subsequent combination and use as a new dataset or as an addition to the original dataset turned out to be effective.

The best kNN classifier, C8, was developed based on the original dataset augmented with a feature based on entropy approximation AE. The best SVM classifier, C7, was developed based on the original dataset augmented with features based on the UMAP algorithm and entropy approximation AE. The average values of metric *MacroF*₁–*score* used to assess the quality of classifiers during cross-validation increased by 16.138% and 4.219%, respectively, compared to the average values of this metric in the case when an unbalanced original dataset was used in the development of classifiers of the same name. The average values of metric *MacroF*₁–*score* increased by 3.660% and 0.440%, respectively, compared to the average values of this metric in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the case when a balanced original dataset was used in the development of the same name.

One can assume that applying the population-based optimization algorithms to search for optimal parameter values of the UMAP algorithm and optimal parameter values of classifiers [58,62,63], working with different formulas for calculating the loss function in the UMAP algorithm [32], entropy and fractal dimension, as well as ideas of hybrid classifiers [34] can ultimately improve the quality of data classification.

6. Conclusions

During this research, we proposed a new approach to the development of datasets used in the development of classifiers in the task of classifying ODs based on blood protein markers. It was suggested to use the results from applying the UMAP dimensionality reduction algorithm to the original dataset and the results of calculating the approximation entropy AE and two fractal dimensions KFD and HFD as new features. In some cases, new features can provide an improvement in the quality of classification with different combinations between themselves or with the original dataset.

The goal of further research is to analyze the prospects for the development and application of cost-effective algorithms in the development of classifiers in the problem of classifying ODs based on blood protein markers. In addition, we plan to study the possibilities of improving the quality of classification by using population optimization algorithms for the values of parameters of the UMAP algorithms and classifier parameters, and also to work with various formulas for calculating entropy and the fractal dimension.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The author declares no conflict of interest.

References

- Global Health Care Outlook. 2021. Available online: https://www2.deloitte.com/cn/en/pages/life-sciences-and-healthcare/ articles/2021-global-healthcare-outlook.html (accessed on 3 January 2023).
- Li, G.; Hu, J.; Hu, G. Biomarker Studies in Early Detection and Prognosis of Breast Cancer. *Adv. Exp. Med. Biol.* 2017, 1026, 27–39. [CrossRef] [PubMed]
- 3. Loke, S.Y.; Lee, A.S.G. The future of blood-based biomarkers for the early detection of breast cancer. *Eur. J. Cancer.* **2018**, *92*, 54–68. [CrossRef] [PubMed]
- 4. Cohen, J.D.; Li, L.; Wang, Y.; Thoburn, C.; Afsari, B.; Danilova, L.; Douville, C.; Javed, A.A.; Wong, F.; Mattox, A.; et al. Detection and localization of surgically resectable cancers with a multi-analyte blood test. *Science* **2018**, *359*, 926–930. [CrossRef]
- 5. Killock, D. CancerSEEK and destroy—a blood test for early cancer detection. Nat. Rev. Clin. Oncol. 2018, 15, 133. [CrossRef]
- Hao, Y.; Jing, X.Y.; Sun, Q. Joint learning sample similarity and correlation representation for cancer survival prediction. BMC Bioinform. 2022, 23, 553. [CrossRef] [PubMed]
- 7. Núñez, C. Blood-based protein biomarkers in breast cancer. Clin. Chim. Acta. 2019, 490, 113–127. [CrossRef] [PubMed]
- Du, Z.; Liu, X.; Wei, X.; Luo, H.; Li, P.; Shi, M.; Guo, B.; Cui, Y.; Su, Z.; Zeng, J.; et al. Quantitative proteomics identifes a plasma multi protein model for detection of hepatocellular carcinoma. *Sci. Rep.* 2020, *10*, 15552. [CrossRef] [PubMed]
- 9. Kalinich, M.; Haber, D.A. Cancer detection: Seeking signals in blood. Science 2018, 359, 866–867. [CrossRef]
- 10. Song, C.; Li, X. Cost-Sensitive KNN Algorithm for Cancer Prediction Based on Entropy Analysis. *Entropy* **2022**, *24*, 253. [CrossRef]
- Huang, S.; Cai, N.; Pacheco, P.P.; Narrandes, S.; Wang, Y.; Xu, W. Applications of Support Vector Machine (SVM) Learning in Cancer Genomics. *Cancer Genom. Proteom.* 2018, 15, 41–51. [CrossRef]
- Sepehri, M.M.; Khavaninzadeh, M.; Rezapour, M.; Teimourpour, B. A data mining approach to fistula surgery failure analysis in hemodialysis patients. In Proceedings of the 2011 18th Iranian Conference of Biomedical Engineering (ICBME), Tehran, Iran, 14–16 December 2011; pp. 15–20. [CrossRef]
- 13. Rezapour, M.; Zadeh, M.K.; Sepehri, M.M. Implementation of Predictive Data Mining Techniques for Identifying Risk Factors of Early AVF Failure in Hemodialysis Patients. *Comput. Math. Methods Med.* **2013**, 2013, 830745. [CrossRef] [PubMed]
- 14. Rezapour, M.; Zadeh, K.M.; Sepehri, M.M.; Alborzi, M. Less primary fistula failure in hypertensive patients. *J. Hum. Hypertens.* **2018**, *32*, 311–318. [CrossRef] [PubMed]

- Toth, R.; Schiffmann, H.; Hube-Magg, C.; Büscheck, F.; Höflmayer, D.; Weidemann, S.; Lebok, P.; Fraune, C.; Minner, S.; Schlomm, T.; et al. Random forest-based modelling to detect biomarkers for prostate cancer progression. *Clin. Epigenet.* 2019, *11*, 148. [CrossRef] [PubMed]
- Savareh, B.A.; Aghdaie, H.A.; Behmanesh, A.; Bashiri, A.; Sadeghi, A.; Zali, M.; Shams, R. A machine learning approach identified a diagnostic model for pancreatic cancer through using circulating microRNA signatures. *Pancreatology* 2020, 20, 1195–1204. [CrossRef]
- 17. Lv, J.; Wang, J.; Shang, X.; Liu, F.; Guo, S. Survival prediction in patients with colon adenocarcinoma via multi-omics data integration using a deep learning algorithm. *Biosci Rep.* **2020**, *40*, BSR20201482. [CrossRef]
- Chaudhary, K.; Poirion, O.B.; Lu, L.; Garmire, L.X. Deep learning-based multi-omics integration robustly predicts survival in liver cancer. *Clin. Cancer Res.* 2018, 24, 1248–1259. [CrossRef]
- 19. Lee, T.Y.; Huang, K.Y.; Chuang, C.H.; Lee, C.Y.; Chang, T.H. Incorporating deep learning and multi-omics autoencoding for analysis of lung adenocarcinoma prognostication. *Comput. Biol.* 2020, *87*, 107277. [CrossRef]
- Qadri, S.F.; Shen, L.; Ahmad, M.; Qadri, S.; Zareen, S.S.; Akbar, M.A. SVseg: Stacked Sparse Autoencoder-Based Patch Classification Modeling for Vertebrae Segmentation. *Mathematics* 2022, 10, 796. [CrossRef]
- Chawla, N.V.; Bowyer, K.W.; Hall, L.O.; Kegelmeyer, W.P. SMOTE: Synthetic Minority Over-sampling Technique. J. Artif. Intell. Res. 2002, 16, 321–357. [CrossRef]
- Han, H.; Wang, W.Y.; Mao, B.H. Borderline-SMOTE: A New Over-Sampling Method in Imbalanced Data Sets Learning. In Advances in Intelligent Computing; ICIC 2005. Lecture Notes in Computer Science, Huang, D.S., Zhang, X.P., Huang, G.B., Eds.; Springer: Berlin, Heidelberg, 2005; Volume 3644, pp. 878–887. [CrossRef]
- 23. Swana, E.F.; Doorsamy, W.; Bokoro, P. Tomek Link and SMOTE Approaches for Machine Fault Classification with an Imbalanced Dataset. *Sensors* **2022**, *22*, 3246. [CrossRef]
- He, H.; Bay, Y.; Garcia, E.A.; Li, S. ADASYN: Adaptive synthetic sampling approach for imbalanced learning. In Proceedings of the 2008 IEEE International Joint Conference on Neural Networks (IEEE World Congress on Computational Intelligence), Hong Kong, 1–8 June 2008; pp. 1322–1328. [CrossRef]
- 25. Tomek, I. Two modifications of CNN. IEEE Trans. Syst. Man Cybern. 1976, 6, 769–772. [CrossRef]
- 26. Candès, E.J.; Li, X.; Ma, Y.; Wright, J. Robust principal component analysis? J. ACM 2011, 58, 1–37. [CrossRef]
- Jolliffe, I.T.; Cadima, J. Principal component analysis: A review and recent developments. *Phil. Trans. R. Soc. A.* 2016, 374, 20150202. [CrossRef] [PubMed]
- 28. van der Maaten, L.; Hinton, G.E. Visualizing Data using t-SNE. J. Mach. Learn. Res. 2008, 9, 2579–2605.
- 29. McInnes, L.; Healy, J.; Melville, J. UMAP: Uniform manifold approximation and projection for dimension reduction. *arXiv* 2018, arXiv:1802.03426.
- 30. Dorrity, M.W.; Saunders, L.M.; Queitsch, C.; Fields, S.; Trapnell, C. Dimensionality reduction by UMAP to visualize physical and genetic interactions. *Nat. Commun.* **2020**, *11*, 1537. [CrossRef]
- Becht, E.; McInnes, L.; Healy, J.; Dutertre, C.A.; Kwok, I.W.H.; Ng, L.G.; Ginhoux, F.; Newell, E.W. Dimensionality reduction for visualizing single-cell data using UMAP. *Nat. Biotechnol.* 2019, 37, 38–44. [CrossRef]
- 32. Demidova, L.A.; Gorchakov, A.V. Fuzzy Information Discrimination Measures and Their Application to Low Dimensional Embedding Construction in the UMAP Algorithm. *J. Imaging* **2022**, *8*, 113. [CrossRef]
- Yu, W.; Liu, T.; Valdez, R.; Gwinn, M.; Khoury, M.J. Application of support vector machine modeling for prediction of common diseases: The case of diabetes and pre-diabetes. *BMC Med. Inform. Decis. Mak.* 2010, 10, 16. [CrossRef]
- 34. Demidova, L.A. Two-stage hybrid data classifiers based on SVM and kNN algorithms. Symmetry 2021, 13, 615. [CrossRef]
- 35. Khan, S.S.; Madden, M.G. One-class classification: Taxonomy of study and review of techniques. *Knowl. Eng. Rev.* 2014, 29, 345–374. [CrossRef]
- Scholkopf, B.; Williamson, R.C.; Smola, A.J.; Shawe-Taylor, J.; Platt, J. Estimating the support of a high-dimensional distribution. *Neural Comput.* 2001, 13, 1443–1471. [CrossRef] [PubMed]
- 37. Liu, F.T.; Ting, K.M.; Zhou, Z.-H. Isolation-Based Anomaly Detection. ACM Trans. Knowl. Discov. Data 2012, 6, 1–39. [CrossRef]
- Zheng, A.; Casari, A. Feature Engineering for Machine Learning: Principles and Techniques for Data Scientists, 1st ed.; O'Reilly Media, Inc.: Sebastopol, CA, USA, 2018; p. 201.
- COSMIC | Catalogue of Somatic Mutations in Cancer. Available online: https://cancer.sanger.ac.uk/cosmic (accessed on 3 January 2023).
- Zanin, M.; Zunino, L.; Rosso, O.A.; Papo, D. Permutation Entropy and Its Main Biomedical and Econophysics Applications: A Review. *Entropy* 2012, 14, 1553–1577. [CrossRef]
- 41. Zhang, A.; Yang, B.; Huang, L. Feature Extraction of EEG Signals Using Power Spectral Entropy. In Proceedings of the International Conference on BioMedical Engineering and Informatics, Sanya, China, 27–30 May 2008; Volume 2, pp. 435–439. [CrossRef]
- 42. Weng, X.; Perry, A.; Maroun, M.; Vuong, L.T. Singular Value Decomposition and Entropy Dimension of Fractals. *arXiv* 2022, arXiv:2211.12338. [CrossRef]
- 43. Pincus, S.M. Approximate entropy as a measure of system complexity. Proc. Natl. Acad. Sci. USA 1991, 88, 2297–2301. [CrossRef]
- 44. Pincus, S.M.; Gladstone, I.M.; Ehrenkranz, R.A. A regularity statistic for medical data analysis. *J. Clin. Monit. Comput.* **1991**, *7*, 335–345. [CrossRef]

- 45. Delgado-Bonal, A.; Marshak, A. Approximate Entropy and Sample Entropy: A Comprehensive Tutorial. *Entropy* **2019**, *21*, 541. [CrossRef]
- 46. Hjorth, B. EEG Analysis Based on Time Domain Properties. *Electroencephalogr. Clin. Neurophysiol.* **1970**, *29*, 306–310. [CrossRef]
- 47. Galvão, F.; Alarcão, S.M.; Fonseca, M.J. Predicting Exact Valence and Arousal Values from EEG. Sensors 2021, 21, 3414. [CrossRef]
- 48. Shi, C.-T. Signal Pattern Recognition Based on Fractal Features and Machine Learning. Appl. Sci. 2018, 8, 1327. [CrossRef]
- 49. Petrosian, A. Kolmogorov Complexity of Finite Sequences and Recognition of Different Preictal EEG Patterns. In Proceedings of the Computer-Based Medical Systems, Lubbock, TX, USA, 9–10 June 1995; pp. 212–217. [CrossRef]
- 50. Katz, M.J. Fractals and the analysis of waveforms. Comput. Biol. Med. 1988, 18, 145–156. [CrossRef] [PubMed]
- 51. Gil, A.; Glavan, V.; Wawrzaszek, A.; Modzelewska, R.; Tomasik, L. Katz Fractal Dimension of Geoelectric Field during Severe Geomagnetic Storms. *Entropy* **2021**, *23*, 1531. [CrossRef]
- 52. Higuchi, T. Approach to an irregular time series on the basis of the fractal theory. *Phys. D Nonlinear Phenom.* **1988**, *31*, 277–283. [CrossRef]
- 53. Hall, P.; Park, B.U.; Samworth, R.J. Choice of neighbor order in nearest-neighbor classification. *Ann. Stat.* **2008**, *36*, 2135–2152. [CrossRef]
- 54. Nigsch, F.; Bender, A.; Van Buuren, B.; Tissen, J.; Nigsch, A.E.; Mitchell, J.B. Melting point prediction employing k-nearest neighbor algorithms and genetic parameter optimization. *J. Chem. Inf. Model.* **2006**, *46*, 2412–2422. [CrossRef]
- 55. Xing, W.; Bei, Y. Medical Health Big Data Classification Based on KNN Classification Algorithm. *IEEE Access* **2020**, *8*, 28808–28819. [CrossRef]
- 56. Mohanty, S.; Mishra, A.; Saxena, A. Medical Data Analysis Using Machine Learning with KNN. In *International Conference on Innovative Computing and Communications*; Gupta, D., Khanna, A., Bhattacharyya, S., Hassanien, A.E., Anand, S., Jaiswal, A., Eds.; Advances in Intelligent Systems and Computing; Springer: Singapore, 2020; Volume 1166. [CrossRef]
- 57. Chapelle, O.; Vapnik, V.; Bousquet, O.; Mukherjee, S. Choosing multiple parameters for support vector machines. *Mach. Learn.* **2002**, *46*, 131–159. [CrossRef]
- 58. Demidova, L.; Nikulchev, E.; Sokolova, Y. Big data classification using the SVM classifiers with the modified particle swarm optimization and the SVM ensembles. *Int. J. Adv. Comput. Sci. Appl.* **2016**, *7*, 294–312. [CrossRef]
- 59. Schober, P.; Vetter, T.R. Logistic Regression in Medical Research. Anesth Analg. 2021, 132, 365–366. [CrossRef]
- Dai, B.; Chen, R.-C.; Zhu, S.-Z.; Zhang, W.-W. Using Random Forest Algorithm for Breast Cancer Diagnosis. In Proceedings of the 2018 International Symposium on Computer, Consumer and Control (IS3C), Taichung, Taiwan, 6–8 December 2018; pp. 449–452. [CrossRef]
- 61. Acharjee, A.; Larkman, J.; Xu, Y.; Cardoso, V.R.; Gkoutos, G.V. A random forest based biomarker discovery and power analysis framework for diagnostics research. *BMC Med. Genom.* **2020**, *13*, 178. [CrossRef] [PubMed]
- 62. Cheng, S.; Liu, B.; Ting, T.O.; Qin, Q.; Shi, Y.; Huang, K. Survey on data science with population-based algorithms. *Big Data Anal.* **2016**, *1*, 3. [CrossRef]
- 63. Demidova, L.A.; Gorchakov, A.V. Application of bioinspired global optimization algorithms to the improvement of the prediction accuracy of compact extreme learning machines. *Russ. Technol. J.* **2022**, *10*, 59–74. [CrossRef]
- 64. Liu, J.-Y.; Jia, B.-B. Combining One-vs-One Decomposition and Instance-Based Learning for Multi-Class Classification. *IEEE Access* 2020, *8*, 197499–197507. [CrossRef]
- 65. Grandini, M.; Bagli, E.; Visani, G. Metrics for Multi-class Classification: An Overview. arXiv 2020, arXiv:2008.05756.
- 66. Haibo, H.; Yunqian, M. *Imbalanced Learning: Foundations, Algorithms, and Applications;* Wiley-IEEE Press: Hoboken, NJ, USA, 2013; p. 216.
- 67. Krawczyk, B. Learning from imbalanced data: Open challenges and future directions. *Prog. Artif. Intell.* 2016, *5*, 221–232. [CrossRef]
- 68. Dong, W.; Moses, C.; Li, K. Efficient k-nearest neighbor graph construction for generic similarity measures. In Proceedings of the 20th International Conference on World Wide Web, Hyderabad, India, 28 March–1 April 2011; pp. 577–586.
- 69. Damrich, S.; Hamprecht, F.A. On UMAP's true loss function. Adv. Neural Inf. Process. Syst. 2021, 34, 12.
- 70. UMAP: Uniform Manifold Approximation and Projection for Dimension Reduction. Available online: https://umap-learn. readthedocs.io/en/latest/_modules/umap/umap_.html (accessed on 4 January 2023).
- 71. Prusty, S.; Patnaik, S.; Dash, S. SKCV: Stratified K-fold cross-validation on ML classifiers for predicting cervical cancer. *Front. Nanotechnol.* **2022**, *4*, 972421. [CrossRef]
- 72. Slamet, W.; Herlambang, B.; Samudi, S. Stratified K-fold cross validation optimization on machine learning for prediction. *Sink. J. Dan Penelit. Tek. Inform.* 2022, *7*, 2407–2414. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.